

# **CADAT 2024**

The First International Conference on Accessible Digital Agriculture Technologies

ISBN: 978-1-68558-324-8

November 17<sup>th</sup> – 21<sup>st</sup>, 2024

Valencia, Spain

## CADAT 2024 Editors

Sandra Sendra, Universitat Politècnica de València, Spain

## **CADAT 2024**

## Forward

The First International Conference on Accessible Digital Agriculture Technologies (CADAT 2024), held between November 17<sup>th</sup>, 2024, and November 21<sup>st</sup>, 2024, in Valencia, Spain, aimed at integrating cutting-edge Artificial Intelligence (AI) and Internet of Things (IoT) technologies into the agricultural sector. This conference focused on harnessing the power of AI and IoT to revolutionize farming practices, enhance productivity, and ensure sustainable food production for a growing global population. CADAT offered a comprehensive platform for researchers, practitioners, industry experts, and policymakers to exchange ideas, showcase innovations, and collaborate towards shaping the future of smart agriculture. Through interdisciplinary discussions and knowledge sharing, CADAT 2024 aimed at driving the transformative change in the agricultural sector towards sustainability, efficiency, and resilience.

We take here the opportunity to warmly thank all the members of the CADAT 2024 technical program committee, as well as all the reviewers. The creation of such a high-quality conference program would not have been possible without their involvement. We also kindly thank all the authors who dedicated much of their time and effort to contribute to CADAT 2024. We truly believe that, thanks to all these efforts, the final conference program consisted of top-quality contributions. We also thank the members of the CADAT 2024 organizing committee for their help in handling the logistics of this event.

We hope that CADAT 2024 was a successful international forum for the exchange of ideas and results between academia and industry for the promotion of progress related to accessible digital agriculture technologies.

#### CADAT 2024 Chairs

#### **CADAT 2024 General Chair**

Sandra Sendra Compte, Universitat Politecnica de Valencia, Spain

#### **CADAT 2024 Steering Committee**

Chase Q. Wu, New Jersey Institute of Technology, USA Khelifa Benahmed, Tahri Mohamed University – Bechar, Algeria Antonio Javier Garcia-Sanchez, Universidad Politécnica de Cartagena, Spain Jaime Lloret Mauri, Universitat Politecnica de Valencia, Spain Zorica Srđević, University of Novi Sad, Serbia Mohammed Ezziyyani, Université Abdelmalek Essaâdi, Morrocco

## CADAT 2024 Committee

### **CADAT 2024 General Chair**

Sandra Sendra Compte, Universitat Politecnica de Valencia, Spain

### **CADAT 2024 Steering Committee**

Chase Q. Wu, New Jersey Institute of Technology, USA Khelifa Benahmed, Tahri Mohamed University – Bechar, Algeria Antonio Javier Garcia-Sanchez, Universidad Politécnica de Cartagena, Spain Jaime Lloret Mauri, Universitat Politecnica de Valencia, Spain Zorica Srđević, University of Novi Sad, Serbia Mohammed Ezziyyani, Université Abdelmalek Essaâdi, Morrocco

## **CADAT 2024 Technical Program Committee**

Khelifa Benahmed, Tahri Mohamed University - Béchar, Algeria Antonella Di Fonzo, Council for Agricultural Research and Economics | Research Centre for Policies and Bioeconomy, Rome, Italy Mohammed Ezziyyani, Université Abdelmalek Essaâdi, Morrocco Laura Garcia, Universidad Politécnica de Cartagena, Spain Antonio Javier Garcia-Sanchez, Universidad Politécnica de Cartagena, Spain Saeid Homayouni, INRS, Centre Eau Terre Environnement (ETE/TENOR/GEO-AI) - Québec City, Canada Satish R. Jondhale, Savitribai Phule Pune University, Pune, India Jaime Lloret Mauri, Universitat Politecnica de Valencia, Spain Vojtěch Merunka, Czech University of Life Sciences in Prague, Czech Republic Simeon Ndoricimpa, Université du Burundi / Higher Institute of Management, Burundi Lorenzo Palazzetti, University of Perugia, Italy V. Pandiyaraju, VIT University, Chennai, India Christina-Ioanna Papadopoulou, University of Western Macedonia, Kozani, Greece Nikolaos Ploskas, University of Western Macedonia, Greece Javier Rocher Morant, FoodUPV, Universitat Politècnica de València, Spain Sandra Sendra Compte, Universitat Politecnica de Valencia, Spain Zorica Srđević, University of Novi Sad, Serbia Chase Q. Wu, New Jersey Institute of Technology, USA

## **Copyright Information**

For your reference, this is the text governing the copyright release for material published by IARIA.

The copyright release is a transfer of publication rights, which allows IARIA and its partners to drive the dissemination of the published material. This allows IARIA to give articles increased visibility via distribution, inclusion in libraries, and arrangements for submission to indexes.

I, the undersigned, declare that the article is original, and that I represent the authors of this article in the copyright release matters. If this work has been done as work-for-hire, I have obtained all necessary clearances to execute a copyright release. I hereby irrevocably transfer exclusive copyright for this material to IARIA. I give IARIA permission or reproduce the work in any media format such as, but not limited to, print, digital, or electronic. I give IARIA permission to distribute the materials without restriction to any institutions or individuals. I give IARIA permission to submit the work for inclusion in article repositories as IARIA sees fit.

I, the undersigned, declare that to the best of my knowledge, the article is does not contain libelous or otherwise unlawful contents or invading the right of privacy or infringing on a proprietary right.

Following the copyright release, any circulated version of the article must bear the copyright notice and any header and footer information that IARIA applies to the published article.

IARIA grants royalty-free permission to the authors to disseminate the work, under the above provisions, for any academic, commercial, or industrial use. IARIA grants royalty-free permission to any individuals or institutions to make the article available electronically, online, or in print.

IARIA acknowledges that rights to any algorithm, process, procedure, apparatus, or articles of manufacture remain with the authors and their employers.

I, the undersigned, understand that IARIA will not be liable, in contract, tort (including, without limitation, negligence), pre-contract or other representations (other than fraudulent misrepresentations) or otherwise in connection with the publication of my work.

Exception to the above is made for work-for-hire performed while employed by the government. In that case, copyright to the material remains with the said government. The rightful owners (authors and government entity) grant unlimited and unrestricted permission to IARIA, IARIA's contractors, and IARIA's partners to further distribute the work.

## **Table of Contents**

Variety Classification by Image Recognition of Grape Leaf and Berry Ryo Sugiura, Mitsunori Ishihara, Jihyun Lim, Chisa Iwasaki, Yu Oishi, Hiroshi Yakushiji, Yoshihiko Kuwana, and Toshiya Yamamoto	1
A Feature-Based Correlation Approach for Analyzing Price Trends of Citrus in Valencia Province (Spain) Roger Arnau, Jose Manuel Calabuig, Nuria Ortigosa, and Luiza Petrosyan	7
Shore Identification and Navigation Route Modification Function for Autonomous Boats Used to Measure the Depth of Irrigation Ponds Sho Nobumoto, Tsuyosi Nakajima, Kota Oshima, and Yutaka Kaizu	11
Real-Time Egg Detection Using Edge Computer Vision Nicholas Hadjisavvas, Nicolas Nicolaou, and Efstathios Stavrakis	16
Beehive Monitoring Based IoT Technologies and AI Benahmed Khelifa, Merbouh Abdeldjalil, Tab Mohamed Abdeessamd, Lorena Parra, Jaime Lloret, and Bourouis Amina	23
A Monitoring System of Crop and Meteorology Parameters by Bringing Together Physical Sensors and Computer Vision Cameras Francisco Heredia-Blaya, Laura Garcia, Antonio-Javier Garcia-Sanchez, Rafael Asorey-Cacheda, and Joan Garcia-Haro	30
Chatbot Integrated Holographic System for Digital Agriculture and Education	36

Kubra Kara, Daniel Matheus Brandao Lent, Alberto Ivars-Palomares, and Jaime Lloret

## Variety Classification by Image Recognition of Grape Leaf and Berry

Ryo Sugiura, Mitsunori Ishihara, Jihyun Lim, Chisa Iwasaki, Yu Oishi Research Center for Agricultural Information Technology National Agriculture and Food Research Organization (NARO) Tsukuba, Ibaraki, Japan e-mail: rsugiura@naro.affrc.go.jp, ishihara.mitsunori@naro.affrc.go.jp, j\_lim@naro.affrc.go.jp, i.chisa@naro.affrc.go.jp, oishi.yu@naro.affrc.go.jp

Abstract— Protection of the rights of plant breeders is essential to preserve the value of new varieties. However, table grapes, for instance, are easy to propagate vegetatively by grafting or cutting, and once they become popular and the cultivation area increases, they face a high risk of unauthorized cultivation and propagation, leading to overseas outflow. To prevent such infringements, effective methods for promptly identifying protected varieties have been desired, and deep learning-based image recognition can be one of the key techniques. The objective of this study is to verify the possibility of identifying grape varieties using images of leaves and berries. The images of leaves, young berries, and mature berries of Shine Muscat and two similar varieties were captured using smartphone cameras, and an image dataset was created to train and test classification models. Image classification models named VGG16, ResNet50, and Vision Transformer (ViT) were applied and redesigned to classify three categories. After training, these models were tested on 51 images of leaves, 174 images of young berries, and 171 images of mature berries. The models achieved classification accuracies of more than 96.1% for leaves, over 99.4% for young berries, and 100% for mature berries. Although additional testing at different sites or in different years will be needed, these results indicate that image recognition techniques can help identify plant varieties toward infringement detection.

Keywords-image recognition; variety classification; grape leaf; grape berry.

#### I. INTRODUCTION

Measures to prevent superior crop varieties from being grown and propagated without permission are necessary, and breeders must receive profits from licensing the varieties. Furthermore, the protection of the rights of plant breeders is essential for preserving the value of new varieties. However, table grapes, for instance, can easily propagate vegetatively by grafting or cutting; once they become popular and their cultivation area increases, they face a high risk of unauthorized cultivation and propagation, leading to overseas outflow.

Several cases of infringements have been observed worldwide. A private company with a breeder's right of a blueberry variety filed a lawsuit in the Federal Court of Australia, alleging that some farmers were growing and Hiroshi Yakushiji, Yoshihiko Kuwana, Toshiya Yamamoto Department of Intellectual Property National Agriculture and Food Research Organization (NARO) Tsukuba, Ibaraki, Japan e-mail: aoe@naro.affrc.go.jp, kuwana@naro.affrc.go.jp, toshiya@naro.affrc.go.jp

selling the variety without any license [1]. Another US company that has several patented grape and cherry varieties launched administrative complaints against the illegal sale of propagation and harvested material of their varieties [2]. An Italian Court admitted the infringement of a breeder's right and stopped an alleged grape producer from harvesting and selling the infringed variety [3]. In Japan, young, highquality grape vines have been cultivated outside the country [4]. These infringements were often confirmed using DeoxyriboNucleic Acid (DNA) profiling or comparison tests. Although DNA analysis and comparison tests are currently among the most reliable ways to identify varieties [5][6][7], they usually take time to obtain results and are sometimes tedious. It might be difficult to develop a method that completely replaces conventional methods at present. However, a simple and prompt complementary method is desired, and Deep Learning (DL)-based image recognition [8][9][10][11][12] can be one of the key techniques. This study proposes an image-based method to identify crop varieties using Deep Neural Networks (DNNs). DNNs are often used for numerous tasks, including image classification, and DL is the easiest and best way to generate classification models.

The objective of this study is to verify the possibility of identifying crop varieties using images. Grapes of the three varieties were selected, and images of their leaves and berries were captured to create the training dataset. Three classification models were trained using this dataset, and their classification accuracies were compared.

The plan of the paper is as follows. In Section II, we mainly describe the details of the image dataset and the DNN-based classification models. Section III shows the classification accuracies obtained by the trained models and discusses the results of visualizing the regions focused by the model for classification. Section IV addresses the concluding remarks.

#### II. MATERIALS AND METHODS

#### A. Image Dataset

The grape varieties used in this study were Shine Muscat [13] and its similar varieties, Muscat of Alexandria and Rosario Bianco. Images of the leaves and berries of these

varieties were captured using three smartphone cameras (SONY Xperia 10IV, Google Pixel 7, Apple iPhone 13 Mini). A single leaf was laid face up on a black plate immediately after sampling off the trees, and an image was captured using one of the cameras. The cameras were held by hands facing down at a distance of approximately 35 cm from the leaves. All leaves were sampled from trees in the experimental field of National Agriculture and Food Research Organization (NARO) in Akitsu (34.33°N, 132.82°E), Hiroshima, Japan, from July 3rd to July 20th in 2023. Similarly, images of young berries sampled from June 20th to June 28th and mature berries sampled from September 21st to September 28th were captured at a distance of approximately 20 cm from the berries.

The image sizes in pixels were  $4000 \times 3000$  (Xperia),  $4080 \times 3072$  (Pixel 7), and  $4032 \times 3023$  (iPhone). The spatial resolutions of leaf and berry images were approximately 0.11 and 0.07 mm, respectively. Image data were created by cropping a square region of  $2500 \times 2500$  pixels with a leaf at the center and  $800 \times 800$  pixels for berries. Consequently, 186 image data of leaves, 665 image data of young berries, and 673 image data of mature berries were obtained. Fig. 1 shows examples of image data for the three varieties.

#### B. Training and Evaluation of Classification Models

The image classification models VGG16 [14], ResNet50 [15], and Vision Transformer (ViT) [16] were applied and redesigned to classify the three categories. VGG16 is one of the simplest and most classical Convolutional Neural Networks (CNNs). It stacks 13 convolutional layers and three fully connected layers. ResNet50 is the most commonly used classifier. It has residual connections across layers in the network, which improve network performance in training process. This model is characterized by fast convergence of the trainable weight despite the large number



Figure 1. Image data examples of three varieties.

of layers. ViT is an image classifier based on Transformer architecture, which has been the de facto standard for natural language processing. It divides an input image into a sequence of patches and examines their contextual relationships with each other using Self-Attention, which contributes to model robustness [17].

The Python programs of these three models and the corresponding pre-trained weights were obtained from TorchVision [18], which is part of the DL framework of the PyTorch project [19]. The output layer of each model was modified to match the classification of the three varieties.

To train and test the models, the dataset mentioned in the previous section was randomly divided into those for training, validation and test as summarized in Table I. The models were trained by fine-tuning the pre-trained weights as the initial values. The weights of each model were updated repeatedly using the training data such that the loss (cross entropy) decreased. The loss on the validation data was also calculated for each update. After 100 iterations (epochs), the weights with the lowest validation loss were selected as the best. During training, accuracy was calculated as the percentage ratio of the number of correct predictions to the total number of samples in each epoch.

After the training process, the best models were tested on 51 images of leaves, 174 images of young berries, and 171 images of mature berries to obtain the final accuracies.

#### C. Visualization of Interest Area

For the VGG models, Gradient-Weighted Class Activation Mapping (Grad-CAM) was applied to show the important portions of an input image for making predictions. Grad-CAM uses the gradient information flowing into the last convolutional layer of a CNN to assign importance values to each neuron for a particular decision of interest [20]. The magnitude of influence on the prediction results can be obtained as a heat map overlaying the input image. This helps us understand where the models focus on the leaf or berry to predict variety. The Grad-CAM program was obtained from the website of [21].

#### III. RESULTS AND DISCUSSION

#### A. Loss and Accuracy Variations in Training Proccess

As an example, Fig. 2 shows the variations in loss and accuracy with an increase in the number of epochs for the training process of VGG16 for leaves. The curves of the training and validation losses gradually converged, and those of the training and validation accuracies quickly reached 1.0

TABLE I. NUMBER OF IMAGE DATA

	Train			Va	lidatio	n	Test		
Leaf	108			)8 27			51		
Lear	37	36	35	9	9	9	17	17	17
Young	401			401 90			174		
berry	135	141	125	30	30	30	58	58	58
Mature	409			93			171		
berry	147	129	133	31	31	31	57	57	57

The three numbers in the bottom row of each cell are of image data of Shine Muscat, Rosario Biance, and Muscat of Alexandria.



Figure 2. Variation in loss and accuracy in the training process of VGG16 with leaf images.

within the early epochs. This implies that the training process was successfully performed, and a model was created to classify an image into three varieties. As the variations in loss and accuracy for both young and mature berries on the other models were similar, their figures were omitted from this report.

#### B. Accuracies on Test Data

The test accuracies of the trained models for leaf classification were obtained and are summarized as confusion matrices, as shown in Fig. 3(a). VGG16 and ResNet50 correctly classified all varieties on the 51 test images with 100% accuracy. However, ViT had some misclassifications and its accuracy across the three varieties was 96.1%. One of the 17 images of Muscat of Alexandria was misclassified as Shine Muscat, resulting in a recall of



(c) Accuracy of each model on 171 mature berry images.

Figure 3. Confusion matrices of test accuracies of variety classification. The vertical and horizontal axes are true and predicted labels, respectively.

94.1%, and one of the 17 image of Shine Muscat was misclassified as Muscat of Alexandria, whereas the images of Rosario Bianco were classified correctly.

The confusion matrices for the young berries are shown in Fig. 3(b). The accuracies of VGG16 and ResNet50 were 100% similar to the leaf classifications, whereas that of ViT was 99.4% on the 174 images. One of the 58 images of Muscat of Alexandria was misclassified as Shine Muscat and its recall was 98.2%. In the case of mature berries, the accuracies were 100% across all varieties for the 171 test images, as shown in Fig. 3(c). The accuracy of ViT for young berries was 99.4%, whereas that for mature berries was 100%. This implies that the growth of berries might make the features of the variety on its shape and texture more distinct. The accuracies of ViT for leaves and young berries were not 100%. A large amount of data is required for the ViT to achieve optimal performance. However, ViT still achieved good results across leaves and berries.

VGG16 and ResNet50 provided perfect classification of the test data. The method of capturing images contributes to obtaining such good results, even with these basic models. Using a black background and capturing images immediately above the subjects made it easy to capture morphological characteristics and might lead to perfect classifications. Assuming that the images of leaves and berries on trees had been captured at several angles or distances in the outside environment and the background was not uniform, as in [22][23], it would have been difficult to obtain such good accuracies.

Although additional testing at different sites or in different years may be required, these results indicate that image recognition techniques can help identify plant varieties for infringement detection.

#### C. Visualization by Grad-CAM

Four heat maps generated from leaf images using the Grad-CAM program were used as examples, as shown in Fig. 4. VGG16 is assumed to have classified the images into Shine Muscat by focusing on the leaf surface (Fig. 4(a)), whereas it seems to observe some leaf tips to classify the



images of Rosario Bianco (Fig. 4(b)). For the images of Muscat of Alexandria, the outline of a leaf appeared to be the focus of attention (Fig. 4(c)). The model for mature berry images may observe the surface of the berries in Rosario Bianco and Muscat of Alexandria (Fig. 5(b) and (c)). However, it possibly focused on the pedicel to recognize the Shine Muscat (Fig. 5(a)).

#### IV. CONCLUSIONS

To examine whether image recognition can be used to distinguish crop varieties, images of the three grape varieties were collected, and classification models were generated. The test results demonstrated that all models were highly accurate. VGG16 and ResNet50 attained test accuracies of 100% for the leaf and berry images. The accuracies of ViT were 96.1% and 99.8% for leaves and young berries, respectively. Grad-CAM clarified that VGG16 focused on different parts of an image depending on the variety. Based on these results, any other DNN-based classification model is most likely to provide high accuracy. In other words, these three types of classification issues may be comparatively easy to handle in DNN-based image recognition. This study showed that the three grape varieties could be classified based on images of both leaves and berries. However, a model that can classify only three varieties is not sufficiently practical for making infringement decisions, and the number of varieties to be recognized should increase. Additional training data must be collected to update and verify the models.

In general, image recognition models that maintain 100% classification accuracy for any unseen image are rare, even though they are trained on a large amount of training data. The models trained in this study are not expected to maintain high accuracy and may produce some misclassifications in unseen images, such as those captured at different sites and in different years. Therefore, distinguishing variety infringements based on images alone is unreliable, and DNA analysis or other conventional methods are still required to obtain a correct judgement. Image recognition techniques should be utilized to screen for suspicious varieties prior to DNA analysis, which may provide an effective method for infringement detection.

In conclusion, this study represents a stepping stone for future development of infringement prevention technology. By increasing the number of cultivars that can be identified with image recognition while maintaining the accuracy, this image-based approach will become more practical, contributing to prompt detection of illegal cultivation and overseas outflow of registered varieties.

#### REFERENCES

- Queensland District Registry, Federal Court of Australia, No: QUD184/2019, Oct. 2019, [Online] Available from: https://www.comcourts.gov.au/file/Federal/P/QUD184/2019/ 3851895/event/30050693/document/1482902 2024.10.22
- [2] AgTech MediaGroup, "IFG Triumphs in Plant Variety Rights Case in China," iGrow News, July 2023, [Online] Available from: https://igrownews.com/ifg-triumphs-in-plant-varietyrights-case-in-china/ 2024.10.22

- [3] C. Collen, "Italian court issues landmark grape ruling," EUROFRUIT, Nov. 2022, [Online] Available from: https://www.fruitnet.com/eurofruit/italian-court-issueslandmark-grape-ruling/247547.article 2024.10.22
- [4] The Yomiuri Shimbun, "Protect Japan's Valuable Agricultural Products through Revised Seed Law," The Japan News, Dec. 2020, [Online] Available from: https://japannews.yomiuri.co.jp/editorial/yomiurieditorial/20201225-157388/ 2024.10.22
- [5] M. Okamoto et al., "A target cultivar-specific identification system based on the chromatographic printed array strip method for eight prominent Japanese citrus cultivars," Breeding Science vol. 73(2), pp. 146–157, April 2023, doi: 10.1270/jsbbs.22065.
- [6] G. Würtenberger, "The Object Of Plant Variety Protection and the Scope Deriving Therefrom," CPVO, Mar. 2017, [Online] Available from: https://cpvo.europa.eu/en/news-andevents/articles/object-plant-variety-protection-and-scopederiving-therefrom 2024.10.22
- [7] Ministry of Agriculture, Forestry and Fisheries Japan, "Plant Variety Protection System of Japan,", April 2022, [Online] Available https://www.hinshu2.maff.go.jp/en/about/leaflet.pdf 2024.10.22
- [8] Y. LeCun and Y. Bengio, "Convolutional Networks for Images, Speech, and Time Series," The handbook of brain theory and neural networks, 1995, [Online] Available from: https://www.researchgate.net/publication/216792820\_Convol utional\_Networks\_for\_Images\_Speech\_and\_Time-Series 2024.10.22
- [9] Y. LeCun, Y. Bengio, and G. Hinton, "Deep Learning," Nature vol. 521(7553), pp. 436-444, May 2015, https://doi.org/10.1038/nature14539
- [10] M. Albahar, "A Survey on Deep Learning and Its Impact on Agriculture: Challenges and Opportunities," vol. 13(3):540, 2023, https://doi.org/10.3390/agriculture13030540
- [11] I. V. Srihith, "A Short Review on Deep Learning in Agriculture," Journal of Advanced Research in Artificial Intelligence & It's Applications vol. 1(3), pp. 23-39, June 2024, https://doi.org/10.5281/zenodo.11612440
- [12] I. Kunduracioglu and I. Pacal, "Advancements in Deep Learning for Accurate Classification of Grape Leaves and Diagnosis of Grape Diseases," Journal of Plant Diseases and Protection vol. 131, pp. 1061–1080 Mar. 2024, https://doi.org/10.1007/s41348-024-00896-z
- [13] K. Shirasawa et al., "De novo Whole-Genome Assembly in an Interspecific Hybrid Table Grape 'Shine Muscat'," DNA Research vol. 29, pp. 1–9, Nov. 2022, https://doi.org/10.1093/dnares/dsac040
- [14] K. Simonyan and A. Zisserman, "Very Deep Convolutional Networks for Large-Scale Image Recognition," arXiv:1409.1556, April 2014, https://doi.org/10.48550/arXiv.1409.1556
- [15] K. He, X. Zhang, S. Ren, and J. Sun, "Deep Residual Learning for Image Recognition," arXiv:1512.03385, Dec. 2015, https://doi.org/10.48550/arXiv.1512.03385
- [16] A. Dosovitskiy et al., "An Image is Worth 16x16 Words: Transformers for Image Recognition at Scale," In International Conference on Learning Representations, Jan. 2021, [Online] Available from: https://openreview.net/pdf?id=YicbFdNTTy 2024.10.22
- [17] D. Zhou, Z. Yu, E. Xie, C. Xiao, A. Anandkumar, J. Feng, and J. M. Alvarez, "Understanding the Robustness in Vision Transformers," arXiv:2204.12451, Nov. 2022, https://doi.org/10.48550/arXiv.2204.12451

- [18] The PyTorch Foundation, TorchVision [Online] Available from: https://pytorch.org/vision/stable/models.html 2024.10.22
- [19] The PyTorch Foundation, PyTorch [Online] Available from: https://pytorch.org/ 2024.10.22
- [20] R. R. Selvaraju, M. Cogswell, A. Das, R. Vadantam, D. Parikh, and D. Batra, "Grad-CAM: Visual Explanations from Deep Networks via Gradient-based Localization," arXiv:1610.02391, Dec. 2019, https://doi.org/10.48550/arXiv.1610.02391
- [21] J. Gildenblat, Introduction: Advanced Explainable AI for computer vision, [Online] Available from: https://jacobgil.github.io/pytorch-gradcambook/introduction.html 2024.10.22
- [22] B. Franczyk et al., "Deep learning for grape variety recognition," Procedia Computer Science vol. 176, pp. 1211– 1220, Oct. 2020, https://doi.org/10.1016/j.procs.2020.09.117
- [23] C. S. Pereira, R. Morais, and M. J. C. S. Reis, "Deep Learning Techniques for Grape Plant Species Identification in Natural Images," Sensor vol. 19(22):4850, Nov. 2019, https://doi.org/10.3390/s19224850

## A Feature-Based Correlation Approach for Analyzing Price Trends of Citrus in Valencia Province (Spain)

R. Arnau <sup>(0)</sup>, J. M. Calabuig <sup>(0)</sup>, N. Ortigosa <sup>(0)</sup> and L. Petrosyan <sup>(0)</sup>

Instituto Universitario de Matemática Pura y Aplicada,

Universitat Politècnica de València, Camino de Vera s/n, 46022,

Valencia, Spain

e-mail: ararnnot@posgrado.upv.es, jmcalabu@mat.upv.es, nuorar@upvnet.upv.es, luipet@inf.upv.es

Abstract—In this paper, we are going to present some work in progress results to study and analyze the price variation among different citric varieties in Valencia province from Comunitat Valenciana region (Spain). A data-driven approach is used to represent each citrus variety and season using 5 features for comparing its prices trends using a correlation analysis. Those findings provide the foundation for implementing clustering algorithms, such as k-Medoids, to classify citrus varieties and seasons based on profitability and market behavior.

Keywords-Citrus; price trends; feature extraction; correlation analysis.

#### I. INTRODUCTION

The agriculture sector plays an important role in achieving multiple Sustainable Development Goals (SDGs). Its impact ranges from eradicating hunger (SDG2) and poverty (SDG1), to protecting the environment (SDG12) and promoting health (SDG3). Thus, the transformation towards sustainable agriculture is essential to achieve these goals [1], [2]. In this sense, the citrus fruit sector is of great importance both in the Valencia Region (Comunitat Valenciana) and in Spain. Indeed, the Comunitat Valenciana is the main citrus fruit producing region in Spain. In 2022/2023, the production was around 67% of Spain's citrus (the main varieties grown being oranges, mandarins, and lemons) with a production of around 2.8 million tonnes. In this period, 1.8 million tonnes were exported (out of 4.3 million tonnes in Spain) with an economic value of around 2 million euros (out of 3.4 million euros in Spain) [3], [4].

On the other hand, data science and Artificial Intelligence are tools that, as in other areas, are already being used in the agriculture sector [5], [6]. In this sense, it is important to highlight that the interpretability of Artificial Intelligence (AI) algorithms in general, and of machine learning algorithms in particular, is crucial when models are to be used for decisionmaking. Thus, interpretability not only increases confidence in the models (and their results) but also their validation. One of the most common and easy-to-interpret clustering algorithms is k-Means. In these algorithms, it is common to use the Euclidean distance because it is easy to understand the proximity of the vectors for the formation of the clusters. However, other distances (such as that associated with the correlation between vectors) and similar algorithms (such as k-Medoids) can be used. Citrus prices can experiment changes across seasons and between seasons, depending on factors such as weather conditions, supply and demand, and production costs, among others, see for instance [7]. In this work, we study the price trends of the citrus fruits in the Valencia province over the period 2015-2022. For this purpose, we identify different varieties of these citrus fruits by means of a vector consisting of 5 features. A correlation analysis of these vectors then allows us to study the trends, which allows us to classify citrus fruits according to their profitability.

In a future work, this first trend analysis will allow us to implement clustering algorithms (such as the aforementioned k-Medoids) based on correlations. This analysis could be a valuable tool for planning, decision-making, risk mitigation and revenue optimization in the agricultural sector. It provides key information that can be used by producers, traders, policy makers, investors and other market actors to improve the efficiency and economic stability of the sector and, as mentioned above, a tool to achieve some of the SDGs.

The rest of the paper is structured as follows. Section II presents the database and the methodology used for this analysis. Obtained results are presented in Section III, while the analysis limitations, discussion and conclusions are drawn in Section IV. Finally, the stages of upcoming research are presented in Section V.

#### II. MATERIALS AND METHODS

The starting point (and usually critical point) in all data science projects is the collection of data. In our case, we use the "Reports of the Valencian Agricultural Sector" (Informes del Sector Agrario Valenciano, in Spanish), which contain temporary information (with annual breakdown) on the agrifood situation in the Comunitat Valenciana since 2015 (actually since 1998). These reports are divided into different chapters containing both meteorological data (Chapter III) and agricultural (Chapter IV) and livestock (Chapter V) statistics, as well as agricultural prices (Chapter VI). We would like to point out here that these reports consist of 20 documents in Excel format for each of the available years, which means that they are not easy to use [8]. Fortunately, a file (in CSV format) containing much of the necessary information can be found in [9]. This file contains weekly prices of agricultural products (not only citrus fruits) since 2017. Together with the files downloaded in [8] the data have been completed to have

information since 2015 (from earlier dates there was missing data). Hence, we have filtered the data (to have only citrus fruits), as well as transformed the dates. So, at the end, we have a database containing 6745 rows corresponding to the weekly price of 9 citrus products (different kinds of Oranges such as, for instance, Navel, Blancas, etc., but also Lemons and Clementines). The 9 products have 43 different varieties. Unfortunately, there were some varieties that were not priced for many weeks (even years) so we decided not to use them and only use the data of 27.

After data curation, several features were extracted in order to form a vector that will characterize each year, so that annual trends are analysed to facilitate subsequent decisions taken by producers.

In this work, we have focused on 5 data features extracted for each analysed year. Even so, we are currently analysing and working on some additional ones so that we can obtain a more custom-made characterization. Since citrus products' season in Valencia region generally starts in September and ends in July, we are not focusing on natural years. From now on, we will refer to the year of the month when the product season started as year indicator.

As each citrus product has a different season length, and we have the data information of prices sampled by weeks, the first feature to be considered is the duration d (in number of weeks) that each specific product season lasted each year. The second and the third features were the minimum (m) and the maximum (M) prices paid each season per kilo of product. Another feature to be considered was the variance, calculated as

$$\sigma^{2} = \frac{\sum_{i=1}^{N} (x_{i} - \mu)^{2}}{N},$$
(1)

where  $x_i$  is the price paid for the product on the *i*-th week of the season and N is the number of weeks that we have data information per season. Finally, we consider the coefficient

$$Q = \frac{M - m}{d},\tag{2}$$

calculated per season, so that a measure of the distribution of the data is also considered.

With the purpose of comparing the tendencies and behaviours of the product prices during each season, we study the correlation of the vectors formed by the extracted features. That is, if  $X = (x_1, x_2, ..., x_5)$  and  $Y = (y_1, y_2, ..., y_5)$ represent a fixed variety and a season each, we compute

$$\operatorname{corr}(X,Y) = \frac{\operatorname{cov}(X,Y)}{\sigma_X \cdot \sigma_Y} = \frac{\sum_{i=1}^5 (x_i - \bar{x}) \cdot (y_i - \bar{y})}{\sqrt{\sum_{i=1}^5 (x_i - \bar{x})^2 \sum_{i=1}^5 (y_i - \bar{y})^2}}.$$
(3)

A correlation value close to 1 indicates a strong linear correlation between observations, while a value close to -1 signifies a linear correlation, but in opposite directions. A value near 0 suggests that the observations are not linearly correlated. This measure, related to the so called cosine similarity, allows us to identify the relation between price tendencies.

#### **III. RESULTS**

When computing the previous features, the 1513 prices observations from [8] and [9] result in a dataframe consisting of 27 varieties of citrus, with its correspondent features for each seasons from 2015-2016 to 2020-2021, since the 2021-2022 data is uncompleted and not used. After removing the observations for which a variety in a concrete season has less than 5 prices recorded, only 19 varieties from Valencia province are considered, each one corresponding to Oranges, Mandarins, and Clementines varieties in a season, and the 5 features extracted. As an example, the first 10 rows can be seen in Table I, where the duration is computed as a portion of a year instead of number of weeks, for convenience.

2015-2016 -	1	0.99	0.67	0.95	-0.22	0.96
2016-2017 -	0.99	1	0.6	0.97	-0.15	0.92
2017-2018 -	0.67	0.6	1	0.68	0.095	0.84
2018-2019 -	0.95	0.97	0.68	1	0.076	0.9
2019-2020 -	-0.22	-0.15	0.095	0.076	1	-0.21
2020-2021 -	0.96	0.92	0.84	0.9	-0.21	1
2015	2016 2016	2017 2017	2018 2018	2019 2015	2020 2025	12021

Figure 1. Autocorrelation of features corresponding to "Clausellina/Okitsu" across seasons.

Some correlation results can be seen in Figures 1 and 2. In Figure 1 we show the correlation of "Clausellina/Okitsu" variety, for which the season 2019/2020 is totally uncorrelated with the other 5 studied, probably due to the COVID-19 pandemic, which altered the prices at the end of the season. Also, 2017-2018 season has a tendency of its own, while the other four seasons are fully correlated with each other. In Figure 2, the relation between different varieties is shown. For example, it could be thought that oranges and clementines would present similar trends among them, but one can see that clementine "Clementina Clemenules" variety has a more similar price tendency to oranges than other clementines varieties, which would be the beforehand expected behaviour.

#### IV. DISCUSSION AND CONCLUSIONS

In this paper, we have presented some work in progress results that show how weekly prices can be analyzed in order to find significant patterns and correlations. This is done across different varieties of citrus in Valencia province in seasons from 2015 to 2021. We have found that each variety can be identified with a numeric vector of 5 features, which allows for the comparison of varieties with different numbers of recorded prices. This vector can be used as a tool to reveal the behavior of several citrus varieties along different seasons and two comparisons of its prices trends are shown.

However, it is necessary to recognize some limitations. Seasonal variations influenced by external factors, such as the COVID-19 pandemic, have affected the consistency of correlations for certain varieties, as exemplified by the 2019/2020 season of the "Clausellina/Okitsu" variety. While this effect is shown in the study, those external disruptions may make it necessary to integrate additional methods, such as time series forecasting, to adapt the study to irregular market conditions. Moreover, correlation-based similarity, while effective in capturing relative trends, may lack the intuitive clarity of Euclidean-based clustering, resulting in a less interpretable methodology.

To our knowledge, there are no similar studies in the context of agricultural price analysis, although clustering studies (comparing the aforementioned k-Medoids) do exist in the field of identification of management zones in precision agriculture [10]

#### V. FUTURE WORK

The vector identification found will be first step to find an AI-based model that can help to classify products in terms of profitability. More specifically, with the feature vector that identifies each product and the correlation metric (actually 1 minus the correlation to make it a dissimilarity function), we can apply clustering techniques such as k-Medoids, which is part of the on-going work. While it is true that this may produce some loss of interpretability, this technique may be useful: (1) for being less sensitive to differences in absolute magnitudes between features and, (2) to obtain a pattern of relationships between variables, rather than in absolute distances (as seems to be our case).

In this future work, other metrics added to the correlation as well as time-series analysis will be considered, including an extension of the database to include more features and products, which would be one limitation of the current study.

#### ACKNOWLEDGMENTS

Authors acknowledge the support of "Ayuda a Primeros Proyectos de Investigación (PAID-06-22), Vicerrectorado de Investigación de la Universitat Politècnica de València (UPV)". The first author was supported by "Programa de Ayudas de Investigación y Desarrollo (PAID-01-21), Universitat Politècnica de València (UPV)".

#### REFERENCES

- G. Hurduzeu, R. L. Pânzaru, D. M. Medelete, A. Ciobanu, and C. Enea, "The development of sustainable agriculture in eu countries and the potential achievement of sustainable development goals specific targets (SDG 2)," *Sustainability*, vol. 14, no. 23, p. 15798, 2022, ISSN: 2071-1050. DOI: 10. 3390/su142315798.
- [2] C. M. Viana, D. Freire, P. Abrantes, J. Rocha, and P. Pereira, "Agricultural land systems importance for supporting food security and sustainable development goals: A systematic review," *Science of The Total Environment*, vol. 806, p. 150718, 2022, ISSN: 0048-9697. DOI: https://doi.org/10.1016/j. scitotenv.2021.150718.
- [3] The Valencian Institute for Business Competitiveness Institut Valencià de Competitivitat Empresarial (IVACE), Cítricos de la Comunitat Valenciana, https://www.ivace.es/index.php/es/ component/weblinks/weblink/545-internacional-documentos/ 548-sectores/357-citricos?Itemid=100096&task=weblink.go, 2023. Retrieved: October, 2024.
- [4] Ministry of Agriculture, Fisheries and food Ministerio de Agricultura, Pesca y Alimentación (MAPAMA), Análisis de la Campaña de Cítricos 2023/2024, https://precioscitricos. com/wp-content/uploads/2024/06/informecampana2023 -24citricossept - feb\_tcm30 - 684049.pdf, 2024. Retrieved: October, 2024.
- [5] R. Khan, N. Dhingra, and N. Bhati, "Role of artificial intelligence in agriculture: A comparative study," in *Transforming Management with AI, Big-Data, and IoT*, F. Al-Turjman, S. P. Yadav, M. Kumar, V. Yadav, and T. Stephan, Eds. Cham: Springer International Publishing, 2022, pp. 73–83, ISBN: 978-3-030-86749-2. DOI: 10.1007/978-3-030-86749-2\_4.
- [6] D. Mhlanga, "The Role of FinTech and AI in Agriculture, Towards Eradicating Hunger and Ensuring Food Security," in *FinTech and Artificial Intelligence for Sustainable Development: The Role of Smart Technologies in Achieving Development Goals.* Cham: Springer Nature Switzerland, 2023, pp. 119–143, ISBN: 978-3-031-37776-1. DOI: 10.1007/978-3-031-37776-1\_6.
- [7] R. J. S. Izquierdo, G. García-Martínez, N. Lajara-Camilleri, and G. Orea-Vega, "Price evolution of 'clemenules' and 'navelina' in spain during the period 2007-2012," 2015.
- [8] Valencian Department of Agriculture, Water, Stock and Fisheries - Conselleria de Agricultura, Agua, Ganadería y Pesca, *Informes del Sector Agrario Valenciano*, https://portalagrari. gva.es/es/pye/informes-del-sector-agrario-valenciano, 2024. Retrieved: October, 2024.
- [9] Ministry for the Digital Processing and civil service Ministerio para la transformación digital y de la función pública, *Precios agrarios de origen de la Comunitat Valenciana*, https: //datos.gob.es/es/catalogo/a10002983-precios-agrarios-deorigen-de-la-comunitat-valenciana, 2024. Retrieved: October, 2024.
- [10] A. Gavioli, E. G. de Souza, C. L. Bazzi, K. Schenatto, and N. M. Betzek, "Identification of management zones in precision agriculture: An evaluation of alternative cluster analysis methods," *Biosystems Engineering*, vol. 181, pp. 86–102, 2019, ISSN: 1537-5110. DOI: https://doi.org/10.1016/j. biosystemseng.2019.02.019.

Variety	Season	Duration	m	M	Q	$\sigma^2$
Clausellina/Okitsu	2015-2016	0.12	0.20	0.27	0.61	0.00089
Clausellina/Okitsu	2016-2017	0.15	0.16	0.26	0.65	0.00123
Clausellina/Okitsu	2017-2018	0.13	0.23	0.26	0.22	0.00019
Clausellina/Okitsu	2018-2019	0.21	0.15	0.25	0.47	0.00138
Clausellina/Okitsu	2019-2020	0.87	0.20	0.29	0.10	0.00093
Clausellina/Okitsu	2020-2021	0.10	0.23	0.27	0.42	0.00023
Clementina Arrufatina	2015-2016	0.12	0.27	0.34	0.61	0.00079
Clementina Arrufatina	2016-2017	0.15	0.22	0.38	1.04	0.00189
Clementina Arrufatina	2017-2018	0.21	0.29	0.36	0.33	0.00045
Clementina Arrufatina	2018-2019	0.12	0.23	0.30	0.61	0.00054

TABLE I. FEATURES FROM THE 10 FIRST OBSERVATIONS OF THE DATAFRAME (DETAILS IN SECTION II).

Clausellina/Okitsu -	1.00	0.79	0.62	0.95	0.81	0.49	0.57	0.77	0.47	0.71	0.47	0.59	0.46	0.77	0.55
Clementina Arrufatina -	0.79	1.00	0.41	0.73	0.98	0.59	0.80	0.65	0.36	0.66	0.19	0.57		0.77	0.64
Clementina Clemenules -	0.62	0.41	1.00	0.67	0.41	0.79	0.46	0.83	0.91	0.81	0.88	0.80	0.85	0.81	0.73
Clementina Marisol -	0.95	0.73	0.67	1.00	0.73	0.53	0.49	0.83	0.52	0.67	0.55	0.61	0.50	0.76	0.54
Clementina Oronules -	0.81	0.98	0.41	0.73	1.00	0.56	0.83	0.65		0.69	0.19	0.62		0.77	0.64
Mandarina Clemenvilla -	0.49	0.59	0.79	0.53	0.56	1.00	0.70	0.68	0.78	0.81	0.60	0.84	0.74	0.74	0.82
Mandarina Nadorcott -	0.57	0.80	0.46	0.49	0.83	0.70	1.00	0.48	0.50	0.76	0.21	0.77	0.55	0.76	0.69
Mandarina Ortanique -	0.77	0.65	0.83	0.83	0.65	0.68	0.48	1.00	0.78	0.82	0.75	0.72	0.73	0.81	0.74
Naranja Navel Lane Late -	0.47	0.36	0.91	0.52	0.38	0.78	0.50	0.78	1.00	0.81	0.90	0.88	0.93	0.78	0.79
Naranja Navel Powell -	0.71	0.66	0.81	0.67	0.69	0.81	0.76	0.82	0.81	1.00	0.66	0.85	0.84	0.84	0.84
Naranja Navelina -	0.47	0.19	0.88	0.55	0.19	0.60	0.21	0.75	0.90	0.66	1.00	0.69	0.83	0.63	0.67
Naranja Washington Navel -	0.59	0.57	0.80	0.61	0.62	0.84	0.77	0.72	0.88	0.85	0.69	1.00	0.86	0.85	0.76
Salustiana -	0.46	0.37	0.85	0.50	0.39	0.74	0.55	0.73	0.93	0.84	0.83	0.86	1.00	0.79	0.78
Satsuma Owari -	0.77	0.77	0.81	0.76	0.77	0.74	0.76	0.81	0.78	0.84	0.63	0.85	0.79	1.00	0.77
Valencia Late -	0.55	0.64	0.73	0.54	0.64	0.82	0.69	0.74	0.79	0.84	0.67	0.76	0.78	0.77	1.00
Caselina Cenertina protectiona participanta															

Figure 2. Mean correlation of Orange ("Naranja"), Clementines ("Clementina") and Mandarins ("Mandarina") varieties on the different seasons.

## Shore Identification and Navigation Route Modification Function for Autonomous Boats Used to Measure the Depth of Irrigation Ponds

Sho Nobumoto

Graduate School of Engineering and Science Master's Shibaura Institute of Technology Tokyo, Japan email:al20109@shibaura-it.ac.jp

Kota Oshima Graduate School of Engineering and Science Master's Shibaura Institute of Technology Tokyo, Japan email:ma23039@shibaura-it.ac.jp

Abstract—Many irrigation ponds are at high risk of collapse due to issues such as aging caused by insufficient management and sediment accumulation. The use of autonomous boats to measure water depth and estimate water storage capacity in these ponds has proven to be effective. However, there were some problems, such as discrepancies between the navigational map and actual conditions, which resulted in insufficient depth data collection near the water's edge or the boats running aground. In this study, we developed an autonomous boat system by improving the technology to utilize image recognition for identifying shoreline positions, allowing the boat to safely and adequately approach the water's edge. To achieve this, the boat's structure was modified by flattening the hull to reduce the risk of grounding. More importantly, three essential functions were also implemented: shoreline recognition, distance measurement from the boat, and route modification based on the collected data. These functions were evaluated in an actual irrigation pond, and it was verified that the distance estimation was accurate enough for the boat to safely navigate near the shoreline. However, several issues were identified, including the impact of boat sway and lighting conditions on recognition accuracy, as well as the need for improved recognition of obstacles along the route that are not part of the shoreline.

Keywords-Irrigation ponds; Semantic segmentation; Distance measurement; Autonomous navigation.

#### I. INTRODUCTION

Irrigation ponds, known as "Tameike" in Japan, are artificially constructed reservoirs designed to store water for agricultural purposes. However, due to increasing labor shortages, the maintenance of many of these ponds has become insufficient, leading to their deterioration or, in some cases, complete abandonment. As these ponds age, the associated risks of disasters, including flooding and sediment outflow due to structural failure, have escalated. To mitigate these risks caused by aging irrigation ponds and to ensure Tsuyosi Nakajima College of Engineering Shibaura Institute of Technology Tokyo, Japan email:tsnaka@shibaura-it.ac.jp

Yutaka Kaizu Graduate School of Agricultural and Life Sciences The University of Tokyo Tokyo, Japan email:kaizu@g.ecc.u-tokyo.ac.jp

adequate management in the face of labor shortages, largescale interventions such as dredging or decommissioning may be required. However, informed decision-making on such measures necessitates regular monitoring of the ponds' water storage capacity, along with the accumulation of sediment and its temporal variations.

To address this need, a method for managing water storage by using autonomous boats to comprehensively measure water depth in irrigation ponds was proposed. However, because the route planning was based on map data, accurate depth measurement at the water's edge could not be achieved, resulting in significant estimation errors in the water storage capacity [1].

In this study, we propose a system that builds upon existing technology by modifying the route of the autonomous boat to enable accurate shoreline recognition. This enhancement allows for the collection of sufficient depth data and reducing estimation errors in water storage capacity. To achieve this, the boat's structure was modified by flattening the hull to reduce the risk of grounding. More importantly, three essential functions were also implemented: shoreline recognition, distance measurement from the boat, and route modification based on the collected data. The shoreline recognition function utilizes a semantic segmentation method to determine the shoreline position in camera images, and the distance measurement function employs a stereo matching method to calculate the distance to that position. The route modification function adjusts the boat's route based on the shoreline recognition results to ensure it can approach the shoreline safely and closely. These improvements not only address the issue of insufficient data collection near the water's edge but also mitigate the risk of grounding, ensuring safe and reliable data acquisition.

In this paper, Section 2 discusses previous research and its challenges, Section 3 presents the proposed system, Section 4 details the experiments and results, and Section 5 provides the conclusions.

#### II. PREVIOUS RESEARCH AND ITS CHALLENGES

To conduct high-accuracy depth measurements of irrigation ponds at low cost and within a short period of time, Kaizu et al. [1] developed a small autonomous boat equipped with navigation control and Global Navigation Satellite System (GNSS) sonar. The navigation control system utilizes open-source autopilot software originally designed for drones. With this navigation control, the boat autonomously follows along a route so as not to deviate from the planned navigation route based on the Global Positioning System (GPS) position information.

The boat autonomously navigates along a route set using the Mission Planner software. During navigation, the depth data from the sonar and time data are stored in the flight controller's internal memory. Upon completion of autonomous navigation, the depth data are correlated with GPS information to determine the depth at specific pond locations. Spline interpolation is then applied to all depth measurements along the route to generate a map of the pond bed, enabling the estimation of the pond's capacity.

However, in this study, the navigation route was designed using Mission Planner on a map of the pond obtained from Google Maps. Due to inaccuracies in the map and variations in water levels, significant errors often occurred in the shoreline positions. This led to issues, such as insufficient collection of depth data near the water's edge or the boat running aground.

Figure 1 illustrates the distribution of estimation errors. The black line indicates the route taken by the autonomous boat, while the red, blue, and yellow areas on the water surface represent varying degrees of error. As shown in the figure, the central region allows for relatively accurate depth estimation, whereas the water's edge exhibits significantly larger errors. This discrepancy is due to insufficient depth data collection, which causes deviations from the estimated values due to irregular depth changes in the lakebed near the shoreline.



Figure 1. Distribution of Estimation Errors (reproduced from [1]).

To address this issue, it is essential to accurately determine the shoreline position during the boat's navigation and to adjust the route based on this data. This approach enables the boat to safely approach the shoreline while avoiding the risk of grounding.

#### III. PROPOSED SYSTEM

#### A. Research Objective

The objective of this study is to address the challenges identified in the previous research and propose and evaluate a system that utilizes sensor data to recognize the shoreline and measure the distance to it. This system dynamically adjusts the navigation route to ensure safe and effective autonomous operation of the pond depth measurement boat.

#### B. Structure of the Prototype Boat

To conduct this research, we developed a new prototype boat. Figure 2 shows an image of the prototype boat.



Figure 2. Developed Experimental Boat.

In Figure 2, a bodyboard was used as the hull to facilitate navigation in shallow waters and minimize the risk of grounding. For shoreline recognition, two types of cameras were mounted at the front of the hull. One of these is a duallens 3D stereo synchronized USB camera module (ELP Co., Ltd.), which is used for autonomous navigation and shoreline detection. The other camera serves as a "visual aid" for manual remote control, with its video feed displayed in real-time on the controller's screen via telemetry.

Inside a waterproof case positioned at the center of the hull, the flight controller Pixhawk 6C (Holybro) and the Jetson Nano 2GB (NVIDIA), a development board equipped with a Graphics Processing Unit (GPU) for image processing, are installed. Four ducted fans were mounted on top of the waterproof case. These fans are controlled by the flight controller to manage the boat's movement. Additionally, a sonar system is attached to the rear of the hull to measure the depth of the pond.

The prototype boat performs autonomous navigation based on a route planned using Mission Planner and programmed into the Pixhawk 6C flight controller.

#### C. Functional Configuration

Figure 3 illustrates the functional configuration of the proposed system. The existing system configuration has been enhanced with three additional functions: shoreline recognition, distance measurement, and route modification.

In Figure 3, the camera mounted at the front of the boat periodically captures images. The shoreline recognition function uses the image data to estimate the position of the shoreline and outputs the coordinates of the shoreline within the image. Based on this shoreline position, the distance measurement function uses triangulation to calculate and output the distance to the shoreline. Then, the route modification function dynamically adjusts the navigation route using the measured distance.



Figure 3. Functional Configuration of the Proposed System.

This series of functions is executed by the Jetson Nano 2GB installed on the boat. The Jetson Nano 2GB is connected to the dual-lens 3D stereo synchronized USB camera module, which manages the timing of the image capture and performs the necessary computational processing for each function. The route modification function communicates with the flight controller Pixhawk 6C via a wired connection, enabling real-time navigation commands and control.

#### IV. EXPERIMENT AND RESULTS

The performance of each added function was evaluated through a series of experiments. The purpose, method, and results of these experiments are outlined below.

#### A. Shoreline Recognition Function

#### (1) Purpose and Method

This function represents the initial step of the proposed system, and it is crucial that it can recognize the shoreline in captured images with accuracy comparable to that of a human determining the shoreline through a camera.

To evaluate this accuracy, we tested the inference performance of semantic segmentation[2]. In this experiment, images of shorelines taken from various distances were input into the shoreline recognition function, and performance was assessed. Multiple models with different training parameters, such as the number of epochs and batch sizes, were developed for semantic segmentation, and a comparison was made to select the optimal model.

(2) Results

The system successfully recognized both the water surface and the shoreline with a visually identifiable level of accuracy across various distances.

Figure 4 provides an example of the output from the shoreline recognition function. In this figure, the shoreline is distinctly identified. The image on the right shows how the system classifies the white areas as water (navigable) and the black areas as land (non-navigable).



Figure 4. Example of Output from the Shoreline Recognition Function.

The shoreline recognition function utilizes semantic segmentation. The performance of the trained model was evaluated using a confusion matrix, and metrics including the accuracy, precision, recall, and F1 score [3]. The evaluation results for the trained model are shown in Figure 5. As indicated in Figure 5, accuracy exceeded 90%, suggesting that the data volume and number of training epochs were adequate. Although there were a few cases of misdetection in localized areas, such as reflections, the system successfully classified the water and land and recognize the shoreline in most areas.

Accuracy	0.95415149
Precision	0.92114224
Recall	0.91336736
Specificity	0.969868
F1score	0.91723832

Figure 5. Evaluation Scores.

Additionally, for performance comparison, several models were created with varying combinations of batch size (4, 8) and number of epochs (100, 200). Table 1 presents the accuracy metrics for these different hyperparameter settings.

epochs	batch size	accuracy
100	4	0.956
	8	0.946
200	4	0.972
	8	0.964

From Table 1, it can be observed that, even with the same data volume, models with more epochs achieved higher accuracy, while smaller batch sizes resulted in slightly better accuracy.

However, a challenge of the shoreline recognition function is the presence of obstacles other than the shoreline. Obstacles such as fences, plants, and rocks can hinder the boat's movement, and thus must be detected and avoided during autonomous navigation. Since the method in this study focuses specifically on shoreline recognition, it is unable to detect such obstacles. To address this issue, it is

necessary to develop a function to detect obstacles separately from the shoreline and relay this information to the route modification function.

#### B. Distance measurement function

#### (1) Purpose and Method

The distance measurement function calculates the distance from the boat's current position to the coordinates of the shoreline identified by the shoreline recognition function using the stereo matching method [4]. For the boat to adjust its route effectively while navigating, not only is measurement accuracy important, but real-time computation is also required. Additionally, since the boat needs time to alter its course, it is necessary to measure the distance from a point far enough to allow for this adjustment. Therefore, it is important to understand how the distance to the shoreline affects the accuracy of the distance estimation.

To evaluate this, the accuracy of the distance measurement function is assessed by calculating the difference between the measured distance and the actual distance as an error, and then dividing this error by the actual distance.

(2) Results

Figure 6 presents the results of the experiment. The data demonstrate that the distance measurement function accurately measured distances up to 10 meters with an error margin of less than 5% when the boat was stationary. Given that the distance to the shoreline when the boat initiates deceleration is 5 meters, and the error at this distance was less than 0.1 meters (within 1%), the function is confirmed to be sufficiently accurate for practical applications.



#### Figure 6. Experiment Results.

However, the camera used in this study had issues when the boat was shaking, causing the captured images to blur and making it impossible to detect distances in some cases. Therefore, for the distance measurement function to operate properly, it is necessary to use a camera capable of capturing clearer images in cases of significant shaking caused by wind, for example.

#### C. Route Adjustment Function

#### (1) Purpose and Method

The route modification function prioritizes collision avoidance and the prevention of grounding to ensure safety before making any changes to the navigation route. In this experiment, we evaluated the boat's ability to halt its movement upon approaching the shoreline and to successfully modify its course afterward.

To achieve this, we tested whether the boat could safely navigate while modifying its course.

## (2) Results

Figure 7 illustrates the pre-configured route (yellow line) set in Mission Planner, as well as the actual route followed by the boat (red line).



Figure 7. Pre-configured route and the actual route followed by the boat.

The results are as follows:

The boat successfully altered its course to navigate safely before reaching the shoreline. During the experiment, the boat approached the shoreline 25 times along the pre-configured route, but in 8 instances, the route modification function did not perform adequately. The details are as follows:

- At points 2 and 5, the boat ran aground and became inoperable.
- At points 3, 7, 11, 15, 19, and 23, the boat collided with obstacles, requiring manual operation.

The causes of these route adjustment failures are as follows:

- Unclear images due to boat shaking: 4 incidents.
- Unclear images in shaded areas: 3 incidents.
- Collision with obstacles such as fences: 1 incident, where the shoreline recognition function detected these as shorelines.

To address these issues, the following improvements to the shoreline recognition and distance measurement functions are recommended:

- Shoreline recognition function: Enhance the ability to detect obstacles in addition to the shoreline.
- Distance measurement function: Use of higherresolution cameras to estimate distances with consistent accuracy, regardless of lighting conditions or boat movement.

#### V. CONCLUSION

To address the issue of depth measurement errors due to insufficient data in the waterfront areas, we developed a new prototype boat and incorporated three functions to it: shoreline recognition, distance measurement, and route modification. Through experiments, we verified the effectiveness of these functions. The shoreline recognition function achieved over 90% accuracy in its evaluation metrics, the distance measurement function maintained an error of less than 4.5% at a distance of 10 meters, and the route modification function successfully avoided collisions in two-thirds of the cases.

Moving forward, it is crucial to enhance the functionality and accuracy of each component to ensure the system's practicality and facilitate its application to real-world pond depth measurements.

#### ACKNOWLEDGMENT

This research was supported by JSPS KAKENHI grants 23K11052 and 24K06314.

#### REFERENCES

- M. Kimura, Y. Kaizu, K. Yasuda, and S. Watanabe, "Development and verification of a simple pond depth measurement method using a small RC boat with autopilot control and sonar, Journal of Agricultural Engineering", IDRE Journal, no. 311 (88-2), pp. IV\_17-IV\_19, 2020.
- [2] K. Mukai, N. Hara, and K. Konishi, "A Study on Obstacle Detection from Monocular Camera Images by Semantic Segmentation in a Small Unmanned Vessel, Transactions of the Institute of Systems", Control and Information Engineers, vol. 34, no. 12, pp. 319-321, 2021.
- [3] Y. Nakatani and K. Kakehi, "Improvement of continuous detection method for river scum using U-Net". Journal of Civil Engineering, Series B1 (Hydraulic Engineering). vol. 77, no. 2, I\_895-I\_900, 2021.
- [4] R. Takase, T. Nishi, T. Yoshimi, and Y. Kawai, "Stereo matching based on pixel-wise global tree", IPSJ Research Report. vol. 2015-CVIM-197, 2015.

## **Real-Time Egg Detection Using Edge Computer Vision**

Nicholas Hadjisavvas *Algolysis Ltd* Nicosia, Cyprus nicholas.hadjisavvas@algolysis.com Nicolas Nicolaou Algolysis Ltd Nicosia, Cyprus nicolas@algolysis.com

Efstathios Stavrakis *Algolysis Ltd* Nicosia, Cyprus stathis@algolysis.com

Abstract—The adoption of Artificial Intelligence (AI) in agriculture and animal husbandry has accelerated in recent years, driven by the versatility and relatively low costs for development and deployment of smart systems. However, many farms still rely on aging equipment and manual labour rendering these innovations inapplicable. In turn, the inability to harness AI and modernise operations may pose an existential risk. To address this challenge, we advocate for retrofitting existing machinery with AI-based modules as a practical alternative. In this paper, we demonstrate how a poultry egg grading machine can be enhanced with smart capabilities through the integration of deep learning and low-cost commodity edge hardware to enable precise egg counting. We present the methodology and algorithms behind this system that enables real-time processing while maintaining high accuracy. In a limited set of experiments, we demonstrated that the Raspberry Pi 5 (RPi5) running the EfficientDet-lite0 model performed just as well as a desktop with an NVIDIA GPU (graphics processing unit), accurately counting all the eggs it was presented with.

Keywords-egg counting; smart retrofitting; deep learning.

#### I. INTRODUCTION

Egg production remains to this day as one of the most important farming enterprises providing a steady supply of a highly nutritious and affordable food source. Poultry egg producers are required to follow specific processes for egg handling, processing, labeling, and marketing to ensure the safety and quality of eggs reaching consumers. These are labour-intensive processes that are assisted by purpose-built equipment, such as egg grading and sorting machines, packaging, storage, etc. However, replacing existing equipment and processes can be costly, time consuming and may cause operational disruption. To reduce the cost of modernising existing poultry egg production facilities with little to no interruption the idea of retrofitting can bring about significant gains [1], [2]. Instead of replacing the equipment farmers have learned to rely on, add-on digital devices can be introduced that provide new advanced capabilities.

This work explores the feasibility of "smart retrofitting" for animal farm equipment. We design, implement, and deploy a low-cost, AI-based edge computing system that integrates with a traditional egg grading and sorting machine. The system automatically counts eggs using computer vision and classifies them based on the configuration of the egg sorting machine. This low-cost solution primarily benefits farms seeking to digitally transform on a limited budget (i.e., the proposed Raspberry Pi 5 system costs approximately one hundred euros).

AI-powered computer vision systems have been widely

adopted in animal and food production industries, improving efficiency, product quality, and distribution speed [3], [4]. AI is expected to continue playing a key role in the agrifood industry's transformation [5], [6]. This success is largely due to the availability of pre-trained computer vision models. However, these models usually perform poorly for specialized field applications, such as egg detection, and require to be retrained or fine tuned. Furthermore, most AI models still require substantial computational resources to run in real-time, making them difficult to implement on low-end devices and deploy them in the field.

The system presented in this work demonstrates that with very limited computational resources, widely available AI models can be employed to improve operations in animal farms. Our system provides extremely accurate egg counts through a robust object detection algorithm enabling low-end single-board computers (e.g., the Raspberry Pi) to perform object detection and tracking in real time. The system's hardware is inexpensive (i.e., Raspberry Pi and the Pi Camera) and it can be trivially deployed in the field without expert knowledge.

The remainder of this paper is structured as follows: Section II provides information regarding the setup of the system in the environment that it is intended to be used. Section III provides a brief overview of related research that informed our approach. In Section IV, we outline our methodology and present our solution for egg counting at the edge. Section V details the experiments conducted to evaluate the system and discusses the results. Finally, Section VI provides a summary of the paper and highlights key conclusions.

#### **II. ENVIRONMENT**

The system developed in this work is based on a Raspberry Pi 5 single-board computer with a Pi Camera V2 module. It was deployed to a chicken farm with a Riva-Selegg Grader, configured to sort eggs into four weight classes (Extra Large, Large, Medium, and Small) with a single feeding lane, shown in Figure 1(a). As eggs move along the horizontal feeding lane, they drop into preconfigured collection areas when their weight exceeds the machine's preset value. These areas are slightly inclined, causing the eggs to roll towards the operator, following random paths, until collected by hand. The machine itself is purely mechanical, without digital features for counting or recording data, so egg counting is done manually by the operators.

The vision-based egg counting edge device was placed



Figure 1: (a) A snapshot of the images obtained by our system. Egg weighting positions are below the eggs running horizontally and located above the egg grading zones separated with metal rods aligned vertically. (b) Ceiling-mounted egg counting Raspberry Pi 5 deployed above a Riva Selegg egg grading machine.

directly above and almost perpendicular to the egg grading machine at a distance of around 2m, as shown in Figure 1(b), in order to provide a top view of the feeding lane and the weighting positions of the eggs.

The system is composed of two subsystems running as independent services: (a) a back-end developed in Python and providing for image acquisition, object detection and tracking, egg counting, and (b) a front-end implemented in ReactJS providing a web interface for controlling the system and its parameters (e.g., calibration, start/stop egg detection, view/edit/confirm egg counts, etc.)

Once deployed, the system requires a simple calibration procedure (detailed in Section IV-E) and is then ready for use. Re-calibration is needed if either the egg detector or grading machine are adjusted.

#### III. RELATED WORK

In the past decade, the application of computer vision and AI in agriculture and animal husbandry has seen a significant increase. In poultry egg production, deep learning has been used to address egg grading [7], identify egg defects [8], [9] or assess freshness [10]. Egg counting, alongside detection, has also been a popular use case for deep learning based applications in the industry [11]–[14].

Automated egg counting using deep learning entails the training of a convolutional neural network to detect eggs first and then the development of a tracking and a counting algorithm to maintain eggs detected through time. Thus, we review some relevent work in object detection, tracking, alongside their application in edge computing.

#### A. Object detection Models

Ulaszewski et al. [12] conducted experiments comparing MobileNet-SSD (Single Shot Detector) v2, YOLOv3 (You Only Look Once), and Faster-RCNN (Region-based Convolutional Neural Network) for egg detection and counting on various hardware platforms. Using inference speed (fps) and counting accuracy as the primary performance metrics, their results demonstrated that MobileNet-SSD was the fastest and most reliable model under the specific conditions of their experiments.

Yang et al. [8] employed four different versions of RTMDet (Real-Time Models for object Detection) [15] models to perform egg detection for automated defect observation and sorting. While the focus was on a different egg-related task, all models still conducted pure egg detection. Among these, RTMDet-x demonstrated the highest accuracy, outperforming the other versions.

Subedi et al. [13] tested various YOLO (You Only Look Once) model versions for detecting floor eggs, while Luo et al. [16] enhanced a YOLOv5 model for detecting leaky eggs on a production line, achieving superior performance over YOLOv4 and F-RCNN models. Similarly, Vinod et al. [11] utilized a YOLO model for implementing an egg counting system.

#### B. Tracking

Tracking is a fundamental prerequisite for effective object counting. Tracking algorithms range from simple geometric approaches to more advanced deep learning-based methods, though the latter often come with increased computational demands.

Ulaszewski et al. [12] and Vinod et al. [11] used simple, yet effective, center-based tracking to pair with their detections. Shen et al. [17] also used a similar center location tracking approach to count people in elevators. Other algorithms like SORT [18] and DeepSort [19] are also used for object tracking (e.g Dinh et al. [20] used it for traffic counting). While algorithms such as SORT and DeepSORT are optimized for more robust tracking, it is crucial to account for their increased computational cost, particularly when designing edge applications where processing resources are limited.

#### C. Edge Applications

Maximizing efficiency and performance in real-time object detection and counting remains an open challenge. Chen et al. [21] reviewed the use of Deep Learning with edge computing, highlighting key issues such as latency, scalability, and privacy. They also focused on the challenges of deploying deep learning models on resource-constrained devices, such as the Raspberry Pi.

Tsu-Chuan et al. [17] have worked on a similar task of edge-based people counting in elevators using a MobileNet-SSD object detector and a line of interest counting strategy. They deployed their system on NVIDIA Jetson nano boards. Deployment of counting systems on the edge also has relevant applications in traffic management and monitoring. Duc-Liem Dinh et al. [20] have introduced a low-cost edge-based system utilising object detection for vehicle detecting, tracking and counting.

#### IV. METHODOLOGY

Our methodology involves four steps: (a) acquire real-time time images from a camera observing the egg feeding lane of the egg grading machine, (b) perform object detection inference using a single-shot model, (c) compare detected objects with those of the previous image to track the objects, (d) use a set of predefined zones to count eggs of different grades, that are either defined interactively or automatically via a calibration step.

#### A. Image Acquisition

A simple camera module is used to obtain images in realtime. A standard resolution of 640 x 480 pixels is chosen to provide sufficient image quality for processing.

#### B. Object detection

Single-shot detection models [22]–[24] are regarded as state-of-the-art solutions for real-time object detection. However, achieving real-time performance on edge devices is challenging due to processing power limitations, which render many otherwise effective algorithms impractical for such environments.

The YOLO architecture, depending on the model size chosen, contains a number of parameters in the range of 3.2 to 68.2 million and require 8.7 to 257.8 billion floating point operations (GFLOPs) for a single network forward pass to detect objects in a single image. For reference, the Raspberry Pi 5 can reach around 34 GFLOPS [25], while the Raspberry Pi 4 is rated around 10 GFLOPS. An NVIDIA RTX 3070 Ti discreet GPU can reach 21.75 TFLOPS. It is therefore reasonable to expect that while object detection models can be run on all three hardware configurations, the lower-end Raspberry Pi 4 may have difficulties keeping up with real-time processing. On the other hand, a modern discrete GPU can easily handle larger detection models. This allows for verifying the performance of egg counting algorithms without the risk of reaching a processing power limit.

Since we are only interested in egg detection and counting, it is also necessary to consider the ability of these general purpose detectors to reliably detect eggs. In our tests, we noticed that these models either do not recognize eggs, or they need to be specifically trained with egg samples to be able to perform well.

These limitations highlighted the need for an object detection model with a lighter architecture. Such a model should deliver satisfactory results when trained with an appropriate dataset, without being as computationally intensive as the YOLO models. Lightweight object detection models designed for on-mobile or edge device inference are wellsupported within the open-source community. Examples of these models include MobileNet-SSD [23], TinyYOLO, and EfficientDet [22]. Google's autoML provides a family of object detection models which include some light, mobilesized versions. Probably the smallest model is EfficientDet-Lite0, which offers a good balance between performance and computational efficiency.

Although it has the lowest performance among all the EfficientDet models on the COCO dataset [26], EfficientDet-Lite0 is likely the best fit for our needs due to its lightweight and efficient design. Its documented Mean Average Precision

(mAP) 26.41% [27] reflects its ability to generalize across a dataset with various object classes, many of which are irrelevant to our goal of recognizing just one class of objects (i.e., eggs). Therefore, the model was further trained and finetuned using a curated, custom egg dataset, as described in Section V.

#### C. Object tracking and counting

Efficient object tracking is essential for accurate counting across frames, requiring an algorithm with minimal computational demand and reliable results. Distance-based centroid trackers meet these needs by matching detected objects between frames using simple Euclidean distance calculations. The accuracy of the counting relies on correct tracking, as each detected object is assigned a unique ID to ensure it is only counted once.

Distance-based object tracking methods, such as the centroid tracker, have notable limitations, with their performance heavily influenced by factors like inference frequency (i.e., the number of frames processed per second by the detection model).

Low inference frequency poses a major challenge when running deep learning models on edge devices in real-time. This limitation can negatively affect centroid tracking algorithms, which rely on comparing an object's position between consecutive frames. Processing only a few frames per second while skipping others can degrade the algorithm's performance, as objects may be too far apart in time, leading to unmatched or mismatched objects.

#### D. Counting using Region Of Interest (ROI)

Detection and counting is performed at the feeding lane of the machine. This compartment of the egg grader is responsible for weighing each incoming egg (using weight springs placed along a mechanical conveyor belt at predefined different positions).

First, the feeding lane is divided into four zones, corresponding to extra large (XL), large (L), medium (M) and small (S). Zones are defined as polygonal areas by four points on the image plane, as shown in Figure 2. To count eggs in these zones we adopted a binning approach. We utilise the centroid tracker's results (which include the ID and the centre coordinates of each detected object for each frame) to execute the counting logic. A detected egg is added or removed from the count in any of the bins (zones) based on the location of its center. We use a simple point-in-polygon test to determine if an egg's centroid falls within the region of any given zone. As the eggs move along the feeding lane, eggs are reassigned to zones. As eggs drop from the feeding lane into the collection area of the machine they remain assigned to the last bin they have been detected in.

The pseudocode for the ROI egg counting process is provided in the algorithm shown in Figure 3.

#### E. Calibration: Automatic Zone Computation

The ROI-based counting algorithm relies on the definition of several detection zones. Although this process is a one-off



**Figure 2:** Schematic of the bin counting process using zones. Both eggs with IDs 1 and 2 are assigned to green zone (XL). As they move on the feeding lane (left to right), the eggs are reassigned from the previous bin to the current bin. Therefore, the egg with ID=1 is finally assigned to red bin (S), while the egg with ID=2 is assigned to purple bin (L).

```
// order by which egg grader sorts eggs
Require: zones = [XL,L,M,S]
Require: bins = dict()
while isStreaming do
    frame = camera.getFrame()
    detections = model.detectEggs(frame)
    ids = assignIds(detections)
    for id, centroid in ids do
        for zone in zones do
            if pointInPolygon(centroid, zone.polygon) then
            bins[zone].append(id)
        end if
    end for
    end for
    end while
```

Figure 3: Pseudocode of the ROI-based egg counting algorithm.

procedure once the equipment has been deployed in the field, it is still necessary to provide the system with at least four points per zone. In our system the operator can connect to the edge device with a smartphone or tablet and configure the zones via a web interface interactively. However, this can be time consuming and tedious to perform in an animal farm and therefore we devised an effective automatic identification of the counting zones.

The calibration procedure shown in Figure 4 is as follows:

• At least 12 small eggs are passed through the egg grading machine. The choice of "small" eggs allows eggs to travel across the entire feeding lane.

• For each image, the centroid of each detected egg is extracted using the trained object detector.

• Locations of all centroids on the image plane are accumulated over the entire duration of the calibration procedure, as shown in Figure 4(a).

• During calibration, the grading machine sequentially moves each egg along the top horizontal feeding lane on the weighing springs. Based on the egg's weight, the springs may release it into the appropriate collection area. Eggs spend more time stationary on the weighing springs than in other positions or while rolling into the gathering area, leading to the formation of dense point clusters at these locations. To identify potential counting zones, we apply DBSCAN [28], [29], a density-based clustering algorithm, to group these closely packed points (see Figure 4(b)). The center of each cluster is then calculated by averaging the positions of the points and stored for further use. • The computed centers of the clustered points are fitted to straight lines to identify the actual weighing locations of the egg sorting machines. First, a grayscale filter (Figure 4(c)) is applied to the image plotting all raw object centers, followed by a thresholding operation (Figure 4(d)) that converts the image to binary. This process removes areas with few or no detections. Subsequently, the Hough transform is used on the remaining centroids to extract a set of straight lines.

• These straight lines are then used in conjunction with the previously extracted cluster centers. The centers are fitted to each line. Lines with cluster centers matching the number of weighting positions of the machine are preserved, while the rest are discarded (see Figure 4(f)).

• For the specific egg grader used in this work, there are two weight springs for large and medium eggs each, and one spring for extra large and small eggs. A bounding box for each of the weight spring locations is computed and then the ones which are responsible for the same egg size are merged. The result of this is visualised in Figure 5.

• Masking Optimization: Once the counting zones are computed, a region that encompasses all zones is estimated. That region is a convex hull computed using the corners of all counting zones. Any pixel outside that region act as a mask (i.e., set to 0). This masking is a significant optimization for the egg detector, because no masked pixels are used when egg detecting, significantly improving accuracy and reducing processing.

#### V. SYSTEM EVALUATION & RESULTS

#### A. Experimental Setup

Experiments were conducted to assess the performance of the counting system on three different hardware configurations: (i) a Desktop PC (OS: Windows 10 Pro, CPU: Intel Core i7-4790K, RAM: 32GB, GPU: NVIDIA GeForce RTX 3070 Ti), (ii) a Raspberry Pi 4 (OS: Raspbian, Model: 4B Rev 1.5, SoC: Broadcom BCM2711, Quad core Cortex-A72 (ARM v8) @ 1.8GHz, RAM: 8GB), and (iii) a Raspberry Pi 5 (OS: Raspbian, Model: 5, SoC: Broadcom BCM2712, Quad core Cortex-A76 @ 2.4GHz, RAM: 8GB).

All devices were tested over two different object detection model architectures: (i) the EfficientDet-Lite0 [22], and (ii) the YOLOv8n [24]. Both models were trained with a custom dataset of 2,226 egg image samples (2065 training and 161 validation images). The dataset contains a variety of egg images covering different lighting conditions and angles, as well as different heights between the camera and the eggs. No augmentation was carried out. While the EfficientDet-Lite0 architecture is lightweight enough to run on all three devices, the YOLOv8n model was converted to the NCNN [30] highperformance neural network inference framework optimized



Figure 4: (a) Detection centroids plotted on the XY plane, (b) post application of DBSCAN to the centroid data, with dense locations plotted in red, (c) result after applying grayscale to the plot image, (d, e) result after applying thresholding (add params) and thinning (add params) operations to the image, (f) the final extracted line matching the cluster centers.



**Figure 5:** (a) Left: After step 7 of the calibration algorithm. The locations (red points) of the weighting springs along the conveyor belt are acquired. (b) Middle: During step 8 of the calibration algorithm where each point is enclosed into a polygon. (c) Right: Towards the end of step 8, where the polygons of weight springs that are responsible for the same egg size are merged and the zones are finally formed.

for mobile and embedded platforms.

The methodology was tested over 3 video recordings of the single-lane egg sorting machine in operation, each capturing a counting session of a mixture of 30 to 110 eggs of 4 different egg grades: small (S), medium (M), large (L), and extra large (XL).

#### B. Evaluation Results

For each test case, the system was evaluated for its counting accuracy. Figure 6 presents the results from our experiments. Each row in the figure corresponds to a different video feed, while each plot in a column corresponds to the counts produced by a different device. The blue bar corresponds to the ground truth (i.e., a count obtained by manually counting the eggs), while the orange and green bars correspond to the counts achieved using the EfficientDet-Lite0, and YOLO algorithms respectively.

The results justify our initial hypotheses: the Desktop

Personal Computer (PC) managed to count flawlessly in all scenarios. In particular, on the workstation, arguably the most powerful device in terms of computation capability among all others, the system manages to produce perfectly accurate counts for all videos with the EfficientDet-Lite0 model. On the same device, inference with the YOLO model manages to achieve similar results, only missing the count of a couple of extra large (XL) eggs in Video 1.

The Raspberry Pi 4 counting accuracy suffered from its low compute power, producing the worst results across all devices. The slow processing and thus long inference times, prevented RPi 4 to catch up with the speed of the counting machine and resulted in under-counts in almost all scenarios. In particular, EfficientDet-Lite0 performed inference in  $\approx 0.159$  seconds per frame on RPi 4, while YOLO inference took  $\approx 0.515$  seconds per frame. Worth noting is the fact that our methodology seems to have a positive impact on the egg-counting, since in almost all scenarios the device appears to under- and not over-count the eggs.

Last but not least, the Raspberry Pi 5 appears to draw the processing boundary at least for the performance of EfficientDet-Lite0. Using the EfficientDet-Lite0 model, we managed to achieve perfect counting accuracy for all egg sizes across all three test recordings (see table I) with an inference time of  $\approx 0.046$  seconds per frame. On the other hand, Raspberry Pi 5 with YOLO produces sub-optimal results by missing the count of a considerable amount of eggs, especially those in large (L) and medium (M) groups. With YOLO on Raspberry Pi 5, inference time per frame reached



Figure 6: Produced results for all three devices and models over three different test videos. Each row represents processing of a different video. The first column displays the results achieved on the Desktop PC, the second column the counting performance of the Raspberry Pi 4 and the last column that of Raspberry Pi 5.

 $\approx 0.256$  seconds, 5.6 times greater than the time required by EfficientDet-Lite0 for the processing of an individual frame. In light of the preceding observations and analysis, it is apparent that egg-counting accuracy of our system is influenced by variations in inference frames per second. Additionally, the adaptability of the system to diverse egg grading machines poses a challenge, as modifications to certain components would be required for compatibility across other machine layouts. Future work could focus on transforming the system into a more generalized solution capable of operating on a wider range of equipment configurations. Moreover, extending the evaluation to environments with diverse conditions, such as varied lighting, would provide valuable insights into

the system's robustness. Expanding testing to a larger, more comprehensive dataset would also help assess the system's accuracy on a broader scale, thus enhancing its reliability in practical applications.

## **TABLE I:** ACCURACY OF THE RASPBERRY PI 5 BASED SYSTEM UTILIZING THE EFFICIENTDET-LITE0 MODEL.

Pi5 / EfficientDet-lite0									
	Counted Eggs	Ground truth	Accuracy(%)						
Video 1	60	60	100						
Video 2	117	117	100						
Video 3	30	30	100						

#### VI. CONCLUSION

In this work, we have demonstrated the usage of low-cost, resource-strapped, edge device capable of detection, tracking and counting of eggs. In an attempt to apply the idea of smart retrofitting, we have enhanced existing egg sorting equipment with a small footprint device, a Raspberry Pi 5. Following our methodology, we managed to count a set of graded eggs using visual inspection, in *real time*, and with *very high accuracy*. In fact, in the experiments we conducted, the accuracy of our counting was identical of that of a GPU equipped Desktop PC, and matched the ground truth in all cases. This showcases the advantage of our system compared to the rest in the relevant literature, which is the full capability of our system to operate on edge devices without the availability of computing intensive hardware like GPUs, something that other works in the area relied on.

#### VII. ACKNOWLEDGMENT

This research was funded by the European Union Recovery and Resilience Facility of the NextGenerationEU instrument, through the Research and Innovation Foundation (CODEVELOP-ICT-HEALTH/0322/0061) of the Republic of Cyprus.

#### REFERENCES

- G. Celenta and M. C. De Simone, "Retrofitting techniques for agricultural machines," in *New Technologies, Development and Application III*, I. Karabegović, Ed. Cham: Springer International Publishing, 2020, pp. 388–396.
- [2] D. Jaspert, M. Ebel, A. Eckhardt, and J. Poeppelbuss, "Smart retrofitting in manufacturing: A systematic review," *Journal of Cleaner Production*, vol. 312, pp. 127–555, 2021.
- [3] I. Kutyauripo, M. Rushambwa, and L. Chiwazi, "Artificial intelligence applications in the agrifood sectors," *Journal of Agriculture and Food Research*, vol. 11, p. 100502, 2023.
- [4] I. Kumar, J. Rawat, N. Mohd, and S. Husain, "Opportunities of artificial intelligence and machine learning in the food industry," *Journal of Food Quality*, vol. 2021, p. 1–10, Jul. 2021.
- [5] V.G. Dhanya et al., "Deep learning based computer vision approaches for smart agricultural applications," *Artificial Intelligence in Agriculture*, vol. 6, pp. 211–229, 2022.
- [6] H. Tian, T. Wang, Y. Liu, X. Qiao, and Y. Li, "Computer vision technology in agricultural automation —a review," *Information Processing in Agriculture*, vol. 7, no. 1, pp. 1–19, 2020.
- [7] M. Omid, M. Soltani, M. H. Dehrouyeh, S. S. Mohtasebi, and H. Ahmadi, "An expert egg grading system based on machine vision and artificial intelligence techniques," *Journal of Food Engineering*, vol. 118, no. 1, pp. 70–77, 2013.
- [8] X. Yang, R. B. Bist, S. Subedi, and L. Chai, "A computer vision-based automatic system for egg grading and defect detection," *Animals*, vol. 13, no. 14, 2023.
- [9] M. Turkoglu, "Defective egg detection based on deep features and bidirectional long-short-term-memory," *Computers and Electronics in Agriculture*, vol. 185, p. 106152, 2021.
- [10] E. Nematinia and S. Mehdizadeh, "Assessment of egg freshness by prediction of haugh unit and albumen ph using an artificial neural network," *Journal of Food Measurement and Characterization*, vol. 12, 09 2018.
- [11] A. Vinod, D. Mohanty, A. John, and B. Depuru, "Application of artificial intelligence in poultry farming - advancing efficiency in poultry farming by automating the egg counting using computer vision system," 08 2023.
- [12] M. Ulaszewski, R. Janowski, and A. Janowski, "Application of computer vision to egg detection on a production line in real time." *ELCVIA Electronic Letters on Computer Vision and Image Analysis*, vol. 20, no. 2, p. 113–143, 2 2022.

- [13] S. Subedi, R. Bist, X. Yang, and L. Chai, "Tracking floor eggs with machine vision in cage-free hen houses," *Poultry Science*, vol. 102, no. 6, 2023.
- [14] I. Kanjanasurat, W. Krungseanmuang, V. Chaowalittawin, and B. Purahong, "Egg-counting system using image processing and a website for monitoring," in 2021 7th International Conference on Engineering, Applied Sciences and Technology (ICEAST), 4 2021, pp. 101–104.
- [15] C. Lyu et al., "Rtmdet: An empirical study of designing real-time object detectors," 2022.
- [16] Y. Luo, Y. Huang, Q. Wang, K. Yuan, Z. Zhao, and Y. Li, "An improved yolov5 model: Application to leaky eggs detection," *LWT*, vol. 187, p. 115313, 2023.
- [17] T.-C. Shen and E. T.-H. Chu, "Edge-computing-based people-counting system for elevators using mobilenet-single-stage object detection," *Future Internet*, vol. 15, no. 10, 2023.
- [18] A. Bewley, Z. Ge, L. Ott, F. Ramos, and B. Upcroft, "Simple online and realtime tracking," in 2016 IEEE International Conference on Image Processing (ICIP), 2016, pp. 3464–3468.
- [19] N. Wojke, A. Bewley, and D. Paulus, "Simple online and realtime tracking with a deep association metric," in 2017 IEEE International Conference on Image Processing (ICIP), 2017, pp. 3645–3649.
- [20] D.-L. Dinh, H.-N. Nguyen, H.-T. Thai, and K.-H. Le, "Towards ai-based traffic counting system with edge computing," *Journal of Advanced Transportation*, vol. 2021, p. 1–15, 6 2021.
- [21] J. Chen and X. Ran, "Deep learning with edge computing: A review," *Proceedings of the IEEE*, vol. 107, no. 8, pp. 1655–1674, 2019.
- [22] M. Tan, R. Pang, and Q. V. Le, "Efficientdet: Scalable and efficient object detection," in 2020 IEEE/CVF Conference on Computer Vision and Pattern Recognition (CVPR). Los Alamitos, CA, USA: IEEE Computer Society, 6 2020, pp. 10778–10787.
- [23] W. Liu et al., SSD: Single Shot MultiBox Detector. Springer International Publishing, 2016, p. 21–37.
- [24] J. Redmon, S. K. Divvala, R. B. Girshick, and A. Farhadi, "You only look once: Unified, real-time object detection," *CoRR*, vol. abs/1506.02640, 2015. [Online]. Available: http://arxiv.org/abs/1506. 02640
- [25] V. R. Group, "The gflops/w of the various machines in the vmw research group." [Online]. Available: https://web.eece.maine.edu/ ~vweaver/group/green\_machines.html
- [26] T.-Y. Lin et al., "Microsoft coco: Common objects in context," in *Computer Vision – ECCV 2014*, D. Fleet, T. Pajdla, B. Schiele, and T. Tuytelaars, Eds. Cham: Springer International Publishing, 2014, pp. 740–755.
- [27] AutoML, "Automl implementation of efficientdet," https://github.com/google/automl/tree/master/efficientdet, accessed: 2024-09-10.
- [28] M. Ester, H.-P. Kriegel, J. Sander, and X. Xu, "A density-based algorithm for discovering clusters in large spatial databases with noise," in *Proceedings of the Second International Conference on Knowledge Discovery and Data Mining*, ser. KDD'96. AAAI Press, 1996, p. 226–231.
- [29] E. Schubert, J. Sander, M. Ester, H. P. Kriegel, and X. Xu, "Dbscan revisited, revisited: Why and how you should (still) use dbscan," ACM Trans. Database Syst., vol. 42, no. 3, 7 2017.
- [30] "ncnn high-performance neural network inference computing framework," https://github.com/Tencent/ncnn, accessed: 2024-09-10.

## **Beehive Monitoring Based on IoT Technologies and AI**

Benahmed Khelifa<sup>1</sup>, Merbouh Abdeldjalil<sup>1</sup>, Tab Mohamed Abdeessamd<sup>1</sup>, Lorena Parra<sup>2, 3</sup>, Jaime Lloret<sup>3\*</sup>, Bourouis Amina<sup>1</sup>

<sup>1</sup> Department of Mathematics and Computer Science, Faculty of Exact Sciences, Tahri Mohamed University, Bechar, Algeria <sup>2</sup> Departamento de Producción Agrícola, Escuela Técnica Superior de Ingeniería Agronómica, Alimentaria y de Biosistemas,

Universidad Politécnica de Madrid, 28040, Madrid, Spain

<sup>3</sup> Instituto de Investigación para la Gestión Integrada de Zonas Costeras, Universitat Politècnica de València, 46730 Grau de Gandia, Spain

Emails: benahmed\_khelifa@yahoo.fr, jalilbns08@gmail.com, tab.mr@yahoo.com, lorena.parra@upm.es,

jlloret@dcom.upv.es, amina.bourouis@univ-bechar.dz

Abstract— The bee is the best pollinator of plants, essential to the balance of ecosystems and our food. Beekeeping and honey collection in their traditional forms have existed for decades in southern Algeria. Currently, beekeeping faces many problems: climate change, chemical pesticides, diseases, theft, predators, and pollutants. The aim of this work is to propose a system to monitor the state of a hive from inside and outside to remedy the problems raised. In this article, we propose an automated hive system based on the Internet of things (IoT) and Artificial Intelligence (AI). This system will collect the internal and external climatic parameters of the hive using different wireless sensors and transfer these values to the beekeeper's smartphone. Actuators placed in the hive will regulate the hive climate and perform other actions. The results of the experiment indicate that the proposed system is effective for monitoring and securing a beehive. An accuracy value of 0.9022 indicates that the decision tree model used for monitoring achieves a high level of accuracy.

Keywords- Beehive; Smart beehive; IoT; sensor network; Artificial Intelligence; decision trees.

#### I. INTRODUCTION

Apiculture involves the keeping of honeybees for the purpose of collecting honey and its derivatives. It is an important practice that has many benefits: honey is an essential nutrient which is used in many industries, bees are natural pollinators that heavily influence the biodiversity of our planet and greatly enhance agricultural yield, and investing in this sector represents an opportunity to reduce poverty in rural areas [1][2]. Honey is commonly used as food and medicine, making its protection from contamination crucial to reducing health issues [3].

Bees and their keepers are facing great challenges due to climate change, the use of chemicals in agriculture, pollution, the urbanization of our planet, water scarcity, and overexploitation of plants. For these reasons, several scientists are concerned about the possibility of bee extinction and how it could affect human life in general. Among the issues of apiculture is the sudden hive collapse syndrome where a beehive dies unexpectedly. Another is overheating in the hive due to unprecedented planet-wide temperature increase leading the bees to less food collection and more focus on temperature regulation resulting in less honey production. In addition, sabotage and theft result in incredible loss for beekeepers. Furthermore, natural predators like certain types of birds and insects destroy entire colonies especially when beekeepers keep their hives in distant and rural areas where plant life is more abundant, and pollution is less apparent [4].

To face these problems, beekeepers need to keep constant watch on their hives making sure their temperature is stable, that they receive additional water and food during drought periods, and that they are active and safe from both human and animal enemies. As such, we can say that beekeeper challenges range from hive and bee health, climate, and status monitoring, to controlling their hives for optimal environmental metrics for bee survival and increased yield. Recently, smart IoT and AI technologies are used everywhere in precision agriculture management [5]-[7], including the monitoring and control of environmental metrics.

Considering the needs of beekeepers and the available IoT technologies, we aim in this work to propose a new beehive monitoring system based on both IoT and AI technologies. Our proposed system consists of two subsystems: the first one for monitoring the microclimate of the hive using sensors for temperature, humidity, carbon dioxide (CO2), and weight; the second subsystem is designed to secure the hive against predators, diseases, etc. This security system is based on smart cameras and speakers. Both subsystems use AI processing algorithms, Wifi communications and an Android application ensuring the communication of full system data access by the beekeeper.

The rest of this work is organized as follows. In Section 2, we present a literature review as well as the extraction of parameters/tools used in this research area. In Section 3, we describe the architecture and design of our proposed system. Section 4 presents the results of the work and their discussion. Finally, we conclude this work with a summary and some perspectives in Section 5.

#### II. RELATED WORK

Many approaches have been used throughout the years to leverage technology to assist in collecting data that can be used to study the health and behavior of individual honeybees and their hives.

In [8], the authors proposed an IoT-based beehive monitoring system. This system captures sensor data (temperature, humidity, weight, video and audio) recordings

at the hives and sends them using the Message Queuing Telemetry Transport (MQTT) protocol to a ThingsBoard dashboard. In [9], the authors describe the recent advances in precision beekeeping as systems and as services. In [10], the article proposes a Self-Powered Smart Beehive Monitoring, Control System (SBMaCS) using IoT, and interconnecting various sensors of temperature, humidity, weight, motion s, and flame. In this article, the authors develop a mobile phone application that interacts with the SBMaCS hardware to monitor and control the various parameters related to the beehives. In [11], the authors proposed a smart beehive monitoring by microservices and a Web platform using the following hardware and software: Raspberry Pi, Android OS and Microsoft SQL Server. In [12], smart sensor systems were developed for real-time and long-term measurement of relevant parameters related to beehive conditions such as the hive weight, sounds emitted by the bees, temperature, humidity, and CO2 inside the beehive, as well as weather conditions outside. In [13], an innovative Edge-based IoT solution is presented for the detection of Varroa disease. The solution relies on Tensor Processing Unit (TPU) acceleration for machine learning-based models pre-trained in the hybrid cloud environment for bee identification and Varroa destructor infection detection. This proposed system can detect the presence of varroosis in beehives in real-time with the use of Edge Artificial Intelligence invoked for the analysis of video streams. In [14], the authors propose an IoT based smart beehive monitoring system integrated with advanced sensors, the system monitors temperature, humidity, hive weight, and diurnal cycle. The system provides real-time data, remote connectivity, and actionable insights for beekeepers. The system enables early disease detection, proactive interventions, and optimized hive management.

Based on this literature review, we conclude the following: Most of the research works in the field of smart hive monitoring have been based on various techniques, algorithms, Artificial Intelligence (AI) and IoT technologies. However, none of these works have addressed the issues of hive security or the challenges associated with the arid climate of a Saharan region. Our study aims to integrate these aspects by taking these specific constraints into account, while using Artificial Intelligence for data classification and decision making. This opens new perspectives for sustainable hive management in harsh environments.

#### III. PROPOSED SYSTEM

#### A. System architecture

Figure 1 shows the architecture of our system. The first component is the beehive climate monitoring component. This part is equipped with sensors for temperature, humidity, CO2, pressure, weight and an internal camera. The security component of the beehive includes an external camera, a motion sensor, a proximity sensor, and a GPS. The actuator part contains a fan and a speaker. The software component includes the cloud, databases, and the Web application.



Figure 1. Architecture of the proposed system.

#### B. Description of the used components

In this section we describe the different components included in the proposed system.

- Raspberry Pi

Raspberry Pi is a Controller using the operating system Raspberry Pi OS (Raspbian). This installation necessitates an SD card. To install Raspberry Pi OS on SD card we used Raspberry Pi Imager.

- BME280

The BME280 sensor typically uses the I2C (Inter-Integrated Circuit) protocol for communication with the Raspberry Pi.

DHT11

Installing the Adafruit\_DHT library is a reliable means to acquire temperature and humidity data from DHT series sensors. Connecting the DHT11 sensor to the Raspberry Pi.

- MQ135

Since the Raspberry Pi 3 B+ lacks analog pins, we utilized an Arduino Uno to read the output of the MQ135 sensor. We established a serial communication between the

- HC-SR04

The RPi.GPIO libraries provide convenient functions to access the GPIO pins to interface with HC-SR04 sensor. Connect HC-SR04 to the Raspberry Pi.

Raspberry Pi Camera V2

Installing the Pi camera library in Python allows access and control of the Raspberry Pi Camera Module. Connect Raspberry Pi Camera Module to the Raspberry Pi.

SRD-05VDC-SL-C Relay

The RPi.GPIO libraries provide convenient functions to access the GPIO pins to interface with the SRD-05VDC-SL-C relay. Connect the relay to the Raspberry Pi.

In Figure 2, the experimental components of our monitoring system are identified. Both the pictures of the physical components, (Figure 2 a) to c)), and a scheme of their communication (Figure 2 d)) are displayed.



(a) Beehive External and Internal components.





(b) Weight sensor.



(d) Components communication.

Figure 2. System description including physical components a) to c) and their communication d).

#### C. System funcional modules

Our proposed system performs two primary functions. Firstly, it monitors the beehive by collecting and analyzing data (temperature, humidity, weight, etc.) at regular intervals which is then sent to the cloud for storage and can be accessed through our mobile app. Secondly, it secures the beehive by conducting environmental scans to detect potential hazards (predators, thieves, etc.), with alerts sent to the mobile app. Below, Figure 3 shows the two monitoring modules of a beehive.



Figure 3. Beehive monitoring modules.

The module of beehive climate monitoring system is presented in Figure 4. This system collects temperature, pressure, humidity, CO2, weight, and photos from sensors and a camera at regular intervals.



Figure 4. Climate monitoring diagram.

The collected data is then analyzed by the Decision tree (an AI Algorithm) to identify any necessary actions and send results to the cloud for storage. The data can be easily accessed through the mobile app.

To improve the productivity of the bees as well as their safety from the dangers of the surrounding area, the system in Figure 5 is designed to conduct environmental scans on a regular basis. This involves using a combination of proximity sensors, HC SR501 sensors, and a camera to detect potential hazards in the beehive's surroundings. The proximity sensors can detect any objects that are too close to the hive and may pose a threat to bees like predators or humans. The HC SR501 sensors can detect any movement such as intruders. The camera can capture images of the beehive's environment allowing beekeepers to visually inspect the area for any potential hazards.



Figure 5. Beehive security system.

These used sensors and cameras can detect potential threats to ensure the security of the hive and take appropriate measures. For example, if the proximity sensors detect a predator (bird, beewolf, thief, etc.), the system can warn the beekeeper and send him a notification and a photo. The system through a speaker can trigger a hawk sound to deter nearby birds which effectively protect the hives. The camera can also be used to monitor activity around the hive as well as the detection of diseases such as varroa.

#### IV. RESULTS AND DISCUSSIONS

In this section, we will present the implementation of our beehive monitoring system. The experiment took place in the south-west of Algeria, in Bechar city, which is an arid area with a dry Saharan climate. The data from this experiment are derived from the beehive shown in Figure 2 a).

#### A. Collected data

The data collected in Table 1 are a series of measurements taken at different dates and times, concerning various environmental factors of a beehive. Here is a general description of the data collected from IoT sensors of our proposed system:

1. Date: The dates range from March 8, 2023, to June 10, 2023, capturing a period of several months.

2. Interior Temperature: The temperature recorded inside the location varied between 24.5°C and 37.0°C, with the highest temperature observed on May 29, 2023.

3. Exterior Temperature: The recorded exterior temperature ranged from 27.2°C to 40.4°C, with the highest temperature observed on June 10, 2023.

4. Rainfalls: The data indicates whether rainfall occurred during the measurements.

5. CO2 Levels: The CO2 levels ranged from 465.62 to 538.94 parts per million (ppm), with the highest level observed on May 29, 2023.

6. Humidity: Humidity levels ranged from 32% to 80%, with the highest humidity recorded on April 26, 2023.

7. Weight: The weight measurements indicate a range from 11.38 to 18, representing units specific to the context of the data collection.

8. Pressure: The pressure values ranged from 925.6 Pa to 933.7 Pa.

 TABLE I.
 COLLECTED DATA FORM DIFFERENT SENSORS AND AT DIFFERENT DATES.

Date/Time	Interior temp. (°C)	Exterior temp. (°C)	Rain falls	CO2 (ppm)	Humidity (%)	Wight (Kg)	Pressure (Pa)
08/03/2023 22:38	24.5	27.2	false	490.06	70	15	930.6
26/04/2023 16:53	25.0	28.0	false	498.21	80	18	925.6
27/04/2023 00:55	30.2	31.0	false	506.35	60	12	927.1
29/05/2023 17:07	37.0	38.1	true	538.94	34	11.5	930.1
10/06/2023 15:10	33.0	40.4	false	506.35	32	11.4	933.7

#### B. Decision tree model

In our monitoring system, hive management is done based on AI and Machine Learning (ML) models to improve the efficiency of beekeeping operations. The decision tree is a model created and integrated into our monitoring system for decision making and behive management. Table 2 illustrates the different classes of hive state generated by our decision tree model.

Example			Attri	ibutes		I	Result			
	Humidity	Exterior	Interior	Weight	Co2	Rainfall	Send	Send	Requires	Class
	%RH	Temp.	Temp.	Kg	ppm	in last	Notification	Alert	hive	
		C°	C°			24h			visit	
1	70 -95	9-47	10-36	1-35	440-500	N	N	N	N	Normal
2	70 -95	<8	10-36	1-35	440-500	N	Y	N	N	Hibemation
3	>96	9-47	10-36	1-35	440-500	Ν	Y	Ν	N	Evaporating
										Nectar
4	0 -70	9-35	10-36	1-35	440-500	N	Y	Ν	N	Low
										humidity hive
5	60-85	9-35	10-36	1-35	440-500	N	N	Y	Y	Colony no
										longer in hive
6	70 -95	9-35	10-36	1-35	400-440	N	N	Y	Y	Diminished
										population -
										Reduced C02
										production
7	70 -95	9-47	>38	1-35	440-500	N	N	Y	Y	Hive is too
										hot
8	>96	2-47	10-36	1-35	440-500	Y	N	Y	Y	Hive is too
										damp
9	0-100	9-47	10-36	>'5	440-500	N	Y	Ν	N	Hive is too
										heavy

TABLE II. GENERATED CLASSES BY THE DECISION TREE MODEL

Table 3 shows the evaluation of the performance of the proposed decision tree model system, including the following metrics: F1-Score, recall, and accuracy.

TABLE III. EVALUATION METRICS VALUES

Metrics	Metric value (%)
F1 Score	88.5
Recall	88.3
Accuracy	90.2

These results highlight its excellent performance in classification tasks.

#### C. Discussion and Graphs

Analysis of these graphs provides an insight into the environmental conditions surrounding the hive which can be valuable to beekeepers in managing and understanding hive health and productivity.

From the observations, it has been found that there is a close relationship between the different environmental parameters of the hive (temperature, humidity, CO2, weight, etc.). The ideal conditions for these parameters can be regulated by the bees themselves or with the help of the proposed intelligent system actuators. For example, fans can be used to improve air circulation if the temperature or CO2 becomes too high; a misting system can be used to control humidity if it is insufficient. Another is heating, which can increase the temperature when needed, mainly during cold seasons.

By combining natural bee regulation and advanced technologies, it is possible to create an optimal environment for colony development which can increase productivity and overall hive health. This represents an opportunity for modern beekeeping to evolve towards more sustainable and efficient practices. The gathered data is presented in the next paragraphs. First, the results of temperature inside and outside the beehive on March 10, 2023, are presented in Figures 6 and 7.

It is possible to see that inside the hive the temperatures are lower than outside during the day. Regarding the rest of the monitored parameters inside the hive, we can identify in Figure 8 that the humidity is minimum at midday with maximum values during the night. Finally, the CO2 concentration is maximum in the late hours of the day, as shown in Figure 9, since this is the moment in which the bees are inside the hive.



Figure 6. Variation of Temperature Outside the Hive on March 10, 2023.



Figure 7. Variation of Temperature Inside the Hive on March 10, 2023.



Figure 8. Variation of Relative Humidity Inside the Hive on March 26/27, 2023.



Figure 9. Variation of CO2 Inside the Hive on March 10, 2023.

Moreover, with this gathered data it has been possible to compare the parameters of a beehive on two different dates: March 9/10, 2023, and June 9/10, 2023. This information is presented in the following figures. In Figure 10, we compare the temperature inside the hive. It is possible to see that despite the high difference on date, the temperature inside the hive is very similar. Figure 11 represents the variation on the humidity inside the hive. In this case, there are differences among the dates with high humidity values in March.



Figure 10. Variation of Temperature Inside the Hive in March and June.

In Figure 12, the variation in the beehive weight is depicted. The weight is slightly higher in June than in March.



Figure 11. Variation of Relative Humidity Inside the Hive in March and June.



Figure 12. Variation of Weight Inside the Hive in March and June.

#### V. CONCLUSION AND PERSPECTIVES

This work highlights the growing importance of IoT and AI technologies in monitoring beehives in the Sahara region in southwestern Algeria. These technological advances offer innovative solutions to monitor the health status of the hive and protect the bees from predators such as bee-eating birds, diseases like varroa, and theft, therefore, contributing to the preservation of the ecosystem and the sustainability of beekeeping. The experimental part of this work shows that AI algorithms can analyze data from IoT devices which ensures intelligent monitoring of the hive.

Our future work will provide more details on the results of experiments related to hive security, namely the detection of predators and diseases affecting bee colonies, besides the study of other parameters like bee sounds, and bee behavior based on Artificial Intelligence and Internet of Things technologies. We also plan to evaluate the system in varied environments and integrate more data analysis methods and AI models for better, deeper insights.

#### REFERENCES

- F. Hatjina, G. Mavrofridis, and R. Jones, "Beekeeping in the medditerranean from antiquity to the present," International Symposium. Syros, Nea Moudania, Greece, October 9–11, 2014.
- R. K. Gupta, W. R. Johan, W. V. Veen and A. Gupta," Beekeeping for Poverty Alleviation and Livelihood Security" vol. 1: Technological Aspects of Beekeeping, Springer Science+Business Media Dordrecht, pp. 599–636, 2014.
- [3] M. Wakgari and G. Yigezu,"Honeybee keeping constraints and future prospects," Cogent Food and Agriculture vol. 7, issue 1, pp. 1-31, 2021.

- [4] Nikita, A. Grover, P. Kalia, R. Sinha, and P. Garg, "Colony collapse disorder: A peril to apiculture," Journal of Applied and Natural Science, 14(3), pp. 729-739, ISSN : 0974-9411 (Print), 2231-5209 (Online), 2022.
- [5] O. Debauche et al., "Web monitoring of bee health for researchers and beekeepers based on the internet of things," Procedia computer science, 130, pp. 991-998, 2018.
- [6] D. Boubiche et al., "Advanced industrial wireless sensor networks and intelligent IoT," IEEE Communications Magazine 56 (2), pp. 14-15, 2018.
- [7] I. G. Magarino, R. Muttukrishnan, and J. Lloret, "Human-centric AI for trustworthy IoT systems with explainable multilayer perceptrons," IEEE Access, vol. 7, pp. 125562-125574, 2019.
- [8] R. Tashakkori, A S. Hamza and M. B. Crawford, "Beemon: An IoT-based beehive monitoring system," Computers and Electronics in Agriculture, North Carolina: Science Direct, vol. 190, 2021.
- [9] H. Hadjur, D. Ammar and L. Lefèvre, "Toward an intelligent and efficient beehive: A survey of precision beekeeping systems and services," Computers and Electronics in Agriculture, Elsevier, 192, pp.1-16, 2022.

- [10] E. Ntawuzumunsi, S. Kumaran and L. Sibomana, "Self-Powered Smart Beehive Monitoring and Control System (SBMaCS)," Sensors MDPI, vol. 21, no. 10, 2021.
- [11] S. Aydin and M. N. Aydin, "Design and implementation of a smart beehive and its monitoring system using microservices in the context of IoT and open data,", Computers and Electronics in Agriculture, vol. 196, doi: 10.1016/j.compag.2022.106897, 2022.
- [12] S. Cecchi, S. Spinsante, A. Terenzi and S. Orcioni ,"A Smart Sensor-Based Measurement System for Advanced Bee Hive Monitoring," Sensors Journal, vol. 20, no. 9, 2726; doi:10.3390/s20092726, 2020.
- [13] D. Mrozek, R. Gorny, A. Wachowicz and B. M. Mrozek, "Edge-Based Detection of Varroosis in Beehives with IoT Devices with Embedded and TPU-Accelerated Machine Learning," Appl. Sci. Journal, 11, 11078, 2021.
- [14] A. Dsouza, P. Aditya, and S. Hegde, "HiveLink-IoT based Smart Bee Hive Monitoring System," https://doi.org/10.48550/arXiv.2309.12054, 2023.

## A Monitoring System of Crops and Meteorology Parameters by Bringing Together Physical Sensors and Computer Vision Cameras

Francisco Heredia-Blaya, Laura García, Antonio-Javier Garcia-Sanchez, Rafael Asorey-Cacheda,

Joan Garcia-Haro

Deparment of Information and Communications Technologies Universidad Politécnica de Cartagena (UPCT) Cartagena, Spain

emails: francisco.heredia@edu.upct.es, {laura.garcia, antoniojavier.garcia, rafael.asorey, joang.haro}@upct.es

Abstract—Mediterranean countries have been facing severe droughts in recent years, leading to water restrictions for consumers and farm owners in the most affected areas. This situation has resulted in increased interest for investing in smart systems to monitor crops, aiming to reduce resource usage and improve crop quality. Existing sensor devices typically measure only a limited selection of physical parameters, such as soil moisture, air temperature, and relative humidity, among others. Furthermore, these devices are usually deployed in various parts of the fields and require different power sources. In this paper, we present a system that gathers data not only from physical sensors but also from cameras to implement computer vision models that enable the monitoring of aspects such as insect presence and plant diseases. These tasks are conceived within the AerOS framework. The system consists of a set of electronic devices designed and developed by ourselves, a heterogeneous wireless network for communication among all devices, and a server to manage the network, store the data, process it, and present it to users. All these components are housed in an anti-vandalism case and powered by a single power source, consisting of a solar panel and a battery. This integrated system expands the features of conventional agriculture monitoring systems.

Keywords-agriculture; IoT; monitoring; heterogeneous network.

#### I. INTRODUCTION

The use of sensing devices in agriculture has experienced a significant growth in recent years. It has been fueled by severe droughts and other adverse conditions farmers have suffered, with special dificulties in the Mediterarnean area subject to irrigation water restrictions mandated by local governments. These events have led to the development of multiple Internet of Things (IoT) solutions for agriculture, specifically employing low-cost components [1]. This choice of components is primarly motivated by the limited economic resources that farmers may have. However, this also leads to devices that cannot cope with the weather conditions to which they are exposed for long, or sensors that do not provide reliable readings. Thus, it is necessary to invest in quality materials to ensure the correct performance of sensing devices. Furthermore, these devices are often placed in rural areas without access to the power grid, requiring the provision of alternative sources of power such as batteries and solar panels [2]. However, when a set of devices is deployed, the cost of these materials to power each device increases. Therefore, combining some of these devices

to be powered by the same set of solar panel and battery may alleviate the higher cost of quality components.

New technological solutions aimed at solving agricultural challenges are including not only sensor data, but also data from images that get analized [3]. But deploying cameras on fields is not done in a straightforward manner [4]. Powering the camera is one of the limitations that need to be faced. However, there are suitable solutions on the market that include solar panel powering. The main challenge cameras face is communication. Pictures and images require higher bandwidth than the numerical values (telemetry) obtained from sensors, resulting in the need for high-bandwidth connectivity. 5G can provide high connectivity to agricultural systems, but the cost of endowing every device with 5G connectivity would be prohibitive, as each will require a SIM card module and a contract with a service provider for each SIM card. Thus, new approaches would benefit from adapting to the different wireless communication needs of each device.

Heterogeneous wireless networks consist of several wireless technologies that are used within the same wireless network [5]. A set of gateways enable the information encapsulated in messages to change their format according to the standards of the required wireless communication protocols. This allows cellular technologies such as 4G and 5G, long-range lowbandwidth technologies, such as LoRa (Long Range), and medium-range high-bandwidth technologies such as WiFi to provide wireless connectivity to agricultural smart devices. These types of solutions adapt to different transmission needs, facilitating the deployment of telemetry sensor devices with low bandwidth requirements at far distances from the gateway using LoRa, and high-bandwidth demanding cameras to transmit pictures and videos using WiFi or 5G/4G communication. The flexibility heterogeneous wireless networks provide can be leveraged by solutions that required seamless connectivity to manage systems that operate in the edge-cloud continuum. Meta operating systems such as AerOS [6] are vouching for developing distributed solutions with cross-domain resource orchestration for different type of use cases, including agriculture. Therefore, considering the aforementioned needs, in this paper, we present a system for crop and meteorology monitoring in agricultural fields that include physical sensors and cameras collecting images intended for computer vision solutions. It

comprises a weather-resistant anti-vandalism encapsulation and provides connectivity to heterogeneous wireless devices using a common power source for all devices.

The remainder of the paper is organized as follows. Section 2 reviews the related work. The methodology is detailed in Section 3. Section 4 discusses the results. Lastly, the conclusion and future works are written in Section 5.

#### II. RELATED WORK

Monitoring soil moisture and other soil parameters has been the main focus of precision agriculture for many years. As a result, there have been many works dealing with this topic. Park et al. presented a system that monitored soil moisture, atmospheric temperature, and relative humidity using low-cost sensors and open-source software [7]. The system also included a water pump to control irrigation. Experiments were conducted at a soybean cultivation. The results showed that the system was able to maintain soil moisture at 40%. Chen et al. introduced a high-density in-situ solution for soil moisture monitoring using the Narrow-Band Internet of Things (NB-IoT) protocol [8]. Thus, the devices required a SIM card to communicate. They used an RS485 soil moisture sensor and tested battery life. The results from one year of tests showed better spatialtemporal accuracy of soil moisture, lower cost, and better energy consumption performance using NB-IoT than other technologies such as ZigBee [9].

Favorable meteorological conditions are also paramount for good crop development. However, optimal weather is rare. Therefore, many works focus on monitoring weather parameters to adopt solutions to adverse climatic events. Khan et al. implemented a low-cost solution for weather monitoring [10]. It was comprised of an Arduino Mega, operating as a control system, and sensors for wind speed, wind direction, rain, solar irradiance, and CO2 concentration monitoring. The gathered data was then transmitted to the cloud. Marwa et al. proposed a system that monitored temperature, humidity, and rain in real time using the S-THB-M008 and S-RGB-M002 sensors [11]. Data were sent to a cloud server that included a MySQL database. Their climatic monitoring system was tested in Tunisia for one year. The results revealed the correct performance of the system. Moreover, Rajapaksha et al. designed a handheld weather station for monitoring barometric pressure, air temperature, air humidity, soil moisture, and carbon monoxide [12]. The device uses an ATMEGA 2560 Microcontroller, a power unit with removable and rechargable batteries, an LCD display, a sensor panel, local storage, a micro-USB charging port, and a common port to connect external sensors. The device also allowed for SMS alerts when temperature thresholds were exceeded.

As soil and meteorological aspects are both indispensable in precision agriculture systems, some works have included sensors intended for these two aspects in their solutions. Placidi et al. proposed a Wireless Sensor Network (WSN) for low-cost soil water content monitoring [13]. Their solution included a photoresistor, a temperature sensor, a low-power microcontroller, and a solenoid valve. Tests were performed in Silty Loam and Loamy Sand, with a non-constant sensitivity for the low-cost volumetric water content sensors. Furthermore, the parameters of the non-linear fitting equation were optimized to correlate the analog voltage output to the reference values for the volumetric water content. Singh et al. designed an irrigation system for precision agriculture in urban environments [14]. The system was based on soil and weather conditions including soil moisture, air temperature, relative humidity, wind speed, and wind direction. The system was tested for two months and the data was displayed using the Thingspeak platform.

Lastly, a growing number of works have begun proposing the use of heterogeneous wireless networks in agriculture. Sanjeevi et al. introduced a scalable WSN using devices comprised of air temperature, air humidity, air pressure, and pH sensors, as well as an Arduino board [15]. Tests were performed to determine the performance of a heterogeneous network with WiMax, WiFi, and LTE wireless technologies, showing that each technology can be advantageous depending on the network requirements of the deployment. Furthermore, Jose Agustín Rodríguez-Mejía et al. proposed a heterogeneous WSN for precision agriculture that combined LoRa, Bluetooth Low Energy, Zigbee, and WiFi using low-cost Heltec Wireless Stick V3 devices [16]. The selection of the wireless technology is based on the content to be transmitted and the Received Signal Strength Indicator (RSSI). The tests concluded that their solution was efficient and reliable.

The solution presented in this paper includes soil and weather monitoring as well as a camera and the use of a heterogeneous communication network to transmit the data from the different devices to the servers. This way, we combine the key aspects necessary for up-to-date precision agriculture solutions, and provide a reliable source of data and images to feed Artificial Intelligence models.

#### III. METHODOLOGY

This section describes the implementation of the hardware of the system and the software development.

The system consists of a base station that houses all the sensing devices, batteries, and the solar panel (see Figure 1.) It features an anti-vandalism enclosure with weather-proof materials and the solar panel on top, shielding the inside. Inside, the Suspended Particulate Matter (SPM) sensor, temperature sensor, humidity sensor, wind speed sensor, solar irradiance sensor, soil moisture sensor, and batteries are placed on the different shelves of the housing. The gateway is installed vertically hanging on one of the walls of the enclosure. The camera is placed on the outside of the enclosure to get a full vision. Furthermore, the camera includes its own solar panel for power supply that is also placed on the outside of the enclosure. The additional powering system was engineered to guarantee the operation of the gateway and sensing devices during nighttime and cloudy days.

#### A. Hardware implementation

The sensors included in the base station are distributed into several sensing devices. Namely, a meteorology and soil





Figure 2. System deployed in anti-vandalism enclosure.

Figure 1. Architecture of the system.

monitoring device that includes the air temperature, humidity, wind speed, solar irradiance, and soil moisture sensors. It is important to highlight that the wind sensor requires a higher voltage power supply than the other sensors. Another device with a case that allows air flow through it includes the SPM sensor. Lastly, the camera is deployed on its own.

Figure 2 shows a detailed view of the inside and outside of the enclosure with the different sensing devices. The meteorology and soil monitoring device is composed of the following elements. The Arduino board itself provides a standard 5V power output, which is shared to power all sensors. However, the wind sensor requires a Direct Current (DC) amplifier to step up the voltage from 5 to 12 volts to ensure its proper functioning. Another consideration is the data output of each sensor. Data obtained from these sensors include parameters such as environmental and soil humidity, temperature, wind speed, radiation, and the battery charge level of the device. To access the sampling values of soil moisture, radiation, and battery voltage, the analog pins on the Arduino board are used. Other sensors, such as the temperature and environmental humidity sensor, operate under the Inter-Integrated Circuit (I2C) communication protocol. Therefore, the SDA and SCL pins were used to facilitate the connection, as their output is not directly connected to an analog pin. Lastly, the wind sensor, poses a particular challenge due to the fact that the Arduino board has an internal Analog-to-Digital Converter (ADC) operating in a 0 to 3.3 volts range, while the wind sensor's proportional output operates in a wider range of 0 to 5 volts. To address this mismatch, we have employed an external analog-to-digital converter capable of processing these inputs and generating a compatible output with the I2C protocol. This converter is powered by the Arduino board and uses a pin of the ADC to send data to the board.

The SPM device is made of a suspended particle sensor with a 0.35 to 40  $\mu$ m particle range and a data aquisition frequency ranging from 1 to 30 seconds [17]. The SPM

sensor is connected to the Arduino board through the Serial Peripheral Interface (SPI) interface. The Arduino board has LoRa connectivity and uses this wireless technology to transmit data to the gateway.

The RAK gateway has LoRa, WiFi, and 5G connectivity. The SPM device and the meteorology and soil monitoring device transmit their data through LoRa to the gateway. Conversely, the camera, requiring higher bandwidth, sends images and video to the gateway through WiFi. The gateway is equipped with a SIM card and uses 5G/4G to send the data to the cellular cell of the service provider. If there is no 5G connectivity in the area, other cellular technologies could be used as long as video traffic can be transmitted.

Regarding the power supply for the devices, two distinct sources are used: a solar panel and a battery. The Arduino boards, the gateway, and the battery are connected through a charge controller. While the solar panel generates power, it ensures that the battery is charged and the Arduino boards and gateway are powered. However, when the solar panel does not provide enough power, the battery takes over as the power source for the system until the solar panel starts generating power again or until the battery is depleted. The battery can last up to 5 days in adverse weather conditions with no sunlight. This is the standard for solar panel infrastructures in the area.

#### B. Server implementation

On the server side, the managmenet and processing of the information generated by the sensors to obtain a meaningful and easily interpretable visual representation of the data is programmed. This step is crucial, as it allows better understanding the performance and behavior of the device under different conditions.

The server at the UPCT receives all the data using the LoRaWAN communication protocol. The network server is the Chirp-Stack open-source LoRaWAN server [18]. When the network server receives the messages from the field devices, the acquired data is stored in a database. Due to the different characteristics of the data, including numerical values and images, two types of databases where used. InfluxDB was the

choice for storing and managing time series data [19]. The decision to use InfluxDB is made to its ability to handle large volumes of real-time data and its efficient structure for time series storage, which is well suited for the numerical data gathered from the sensors. Conversely, MongoDB was used to store images, which could not be stored using InfluxDB [20].

Once the databases were configured and running, Grafana [21] was integrated into our system. Grafana allows interactive dashboards with graphs, tables, and other visual elements, making it easier to interpret and analyze the data stored in InfluxDB. Furthermore, Grafana can also embed images and videos retrieved form MongoDB. To achieve the final graphical representation, the necessary queries were made to visualize the extracted data as effectively as possible. Line graphs were chosen to represent the evolution of humidity levels, temperature, wind speed, solar radiation, and soil moisture over time. These line graphs help in identifying trends, seasonal patterns, and significant changes in the sensor data. Additionally, the battery status of the device was presented using pie charts. These charts provide a quick and clear view of the battery charge percentage, which is crucial for evaluating the device's autonomy and energy efficiency. Nevertheless, while line and pie charts were used for our current purposes, Grafana offers a wide range of visualization options. Dashboards can be customized and adapted according to the user specific needs and preferences as well as the requirements of each project.

#### IV. RESULTS

In this section, we present the results from the base station deployed on the fields. These results show some relevant observations from different time periods.

Figure 3 shows the overview of the Grafana dashboard for our system. The user interface was designed so that one can easily view the current values from the sensors. This includes different sizes of particulate matter (PM01, PM2.5, and PM10), wind, solar radiation, humidity, soil moisture, and temperature. Furthermore, the map with the location of the base station and the real-time video from the camera is also displayed on the dashboard.



Figure 3. Grafana dashboard of the system.

The graphs plotting the time-series data for each sensor are also available in another Grafana dashboard. Figure 4 illustrates an extract of temperature data for the variation in temperature experienced at different hours of the day. As it can be seen, nights and early mornings are always the periods with the lowest temperatures, whereas it rises in peak sunlight hours.



Figure 4. Temperature readings.

Humidity readings are represented in Figure 5. The graph indicates that humidity rises at nightime and decresaes through the day up to mid-afternoon. This can vary depending on the time of year.



Figure 5. Humidity readings.

Regarding radiation, it can be observed that it is much higher during sunlight hours. The small variations observed in Figure 6 are due to the clouds creating shadows, as they reduce the brightness and radiation captured by the sensor. Then, light is reduced at night. The fields under monitoring are close to a small town and some light might be appreciated at night.



Figure 6. Solar radiation expressed in Photosynthetic Photon Flux Density (PPFD).

As for the wind sensor, the readings in Figure 7 show that wind is strongest in the afternoon on the field. According to the Beaufort scale, this would correspond to grade 3, which indicates a gentle breeze. It can also be seen that wind readings are more abrupt than other parameters.

Figure 8 depicts the readings for the suspended particulate matter. The particles with a bigger size such as PM10 have higher concentrations. Conversely, the particles with lower



Figure 7. Wind speed readings

sizes, such as PM2.5 and PM1 have smaller concentrations. Particle concentrations on fields can increase due to several reasons. Strong winds, dust storms, or agricultural machinery working can contribute to higher outcomes.

![](_page_38_Figure_4.jpeg)

Figure 8. SPM readings.

The readings regarding soil moisture are represented in Figure 9. It can be observed that soil moisture was decreasing during the night but it increases again after the drip irrigation system is turned on in the morning.

![](_page_38_Figure_7.jpeg)

Figure 9. Soil moisture readings.

Lastly, Figure 10 shows the device's battery level and the incoming voltage. These values help users to monitor potential battery issues during periods of low sunlight.

![](_page_38_Figure_10.jpeg)

Figure 10. Baterry readings.

The images gathered by the camera of our solution are being used for developing computer vision tools for agriculture. The images collected from the fields can be used to feed Artificial Intelligence (AI) engines and detect different aspects from plants, including plant diseases (see Figure 11) or pests [3].

![](_page_38_Picture_14.jpeg)

Figure 11. Output from computer vision plant disease detection feature.

The output from the sensors and computer vision features presented in this section can help farmers improve their irrigation schedules and determine if plants are being affected by a disease, alerting the farmers in case the results are positive. Nonetheless, there are no anomalies in the results presented in this paper.

#### V. CONCLUSION AND FUTURE WORK

The adverse climatic conditions faced by Mediterranean countries have led to the need to implement solutions for monitoring meteorology and soil parameters that affect the crops of these areas. This type of system helps reduce water consumption for irrigation and simultaneously optimize crop growth and production. However, existing approaches typically include only a limited selection of parameters. This paper presents a system for crop and metheorology monitoring intended for precision agriculture. It includes a camera to perform computer vision, increasing the number of features that can be provided to the users. The devices were placed in a custom-made anti-vandalism enclosure. Moreover, a heterogeneous wireless network was deployed to ensure all types of data, with their respective bandwidth requirements, could be transmitted. It included WiFi, LoRa, and 5G connectivity. Finally, the proposed system is powered by renewable energy, so it can be placed anywhere in the field, providing positioning independence.

For future work, this system will be used as part of an agrivoltaic solution to monitor meteorologic and soil parameters, as well as serving as a gateway for other devices deployed to monitor other parts of the fields or greenhouses, and the energy produced by the solar panels of the agrivoltaic system. Furthermore, more computer vision models will be developed to increase the available features and/or detecting options.

#### **ACKNOWLEDGEMENTS**

This work was supported by the grants PID2023-148214OB-C21, PCI2024-153485, TED2021-129336B-I00, and FJC2021-

047073-I funded by MICIU/AEI/10.13039/501100011033 and by the European Union NextGenerationEU/PRTR. This work was also funded by Fundación Séneca (22236/PDC/23). This work was also a result of the ThinkInAzul and AgroAlNext programmes, funded by Ministerio de Ciencia, Innovación y Universidades (MICIU) with funding from European Union NextGenerationEU/PRTR-C17.11 and by Fundación Séneca with funding from Comunidad Autónoma Región de Murcia (CARM). This research was also contextualized to DAIMon, a cascade funding action deriving from the Horizon Europe project aerOS, funded by the European Commission under grant number 101069732.

#### REFERENCES

- F. Viani, M. Bertolli, M. Salucci, and A. Polo, "Low-cost wireless monitoring and decision support for water saving in agriculture", *IEEE sensors journal*, vol. 17, no. 13, pp. 4299– 4309, 2017.
- [2] S. Sadowski and P. Spachos, "Solar-powered smart agricultural monitoring system using internet of things devices", in 2018 IEEE 9th Annual Information Technology, Electronics and Mobile Communication Conference (IEMCON), IEEE, 2018, pp. 18–23.
- [3] M. Vilar-Andreu, L. García, A.-J. Garcia-Sanchez, R. Asorey-Cacheda, and J. Garcia-Haro, "Enhancing precision agriculture pest control: A generalized deep learning approach with yolov8based insect detection", *IEEE Access*, 2024.
- [4] D. Vasisht et al., "Farmbeats: An iot platform for data-driven agriculture", in 14th USENIX Symposium on Networked Systems Design and Implementation (NSDI 17), 2017, pp. 515–529.
- [5] J. Lloret, L. García, J. M. Jimenez, S. Sendra, and P. Lorenz, "Cluster-based communication protocol and architecture for a wastewater purification system intended for irrigation", *IEEE Access*, vol. 9, pp. 142 374–142 389, 2021.
- [11] C. Marwa, S. B. Othman, and H. Sakli, "Iot based low-cost weather station and monitoring system for smart agriculture", in 2020 20th International Conference on Sciences and Techniques of Automatic Control and Computer Engineering (STA), IEEE, 2020, pp. 349–354.
- [12] T. T. Rajapaksha, A. Alexander, L. Fernando, A. Than, and H. L. Nguyen, "Real-time weather monitoring and iot-based palmtop device for smart agriculture", *SN Computer Science*, vol. 3, no. 1, p. 91, 2022.

- [6] Aeros: European iot-edge-cloud, https://aeros-project.eu/, Accessed: 2024-10-28.
- [7] J. K. Park and J. Kim, "Agricultural environment monitoring system to maintain soil moisture using iot", *Journal of Internet* of *Things and Convergence*, vol. 6, no. 3, pp. 45–52, 2020.
- [8] D. Chen, N. Chen, X. Zhang, H. Ma, and Z. Chen, "Nextgeneration soil moisture sensor web: High-density in situ observation over nb-iot", *IEEE Internet of Things Journal*, vol. 8, no. 17, pp. 13 367–13 383, 2021.
- Zigbee specifications, https://zigbeealliance.org/wp-content/ uploads/2019/11/docs-05-3474-21-0csg-zigbee-specification. pdf, Accessed: 2024-11-13.
- [10] A. Khan, M. Makmud, M. Miskon, A. Nair, and K. Bidin, "Development of a weather station using iot platform based arduino integrated with a control system for smart agriculture applications", in 2024 IEEE International Conference on Automatic Control and Intelligent Systems (I2CACIS), IEEE, 2024, pp. 134–138.
- [13] P. Placidi *et al.*, "Monitoring soil and ambient parameters in the iot precision agriculture scenario: An original modeling approach dedicated to low-cost soil water content sensors", *Sensors*, vol. 21, no. 15, p. 5110, 2021.
- [14] D. K. Singh, R. Sobti, A. Jain, P. K. Malik, and D.-N. Le, "Lora based intelligent soil and weather condition monitoring with internet of things for precision agriculture in smart cities", *IET communications*, vol. 16, no. 5, pp. 604–618, 2022.
- [15] P. Sanjeevi *et al.*, "Precision agriculture and farming using internet of things based on wireless sensor network", *Transactions on Emerging Telecommunications Technologies*, vol. 31, no. 12, e3978, 2020.
- [16] J. A. Rodríguez-Mejía, S. Sendra, A. Ivars-Palomares, and J. Lloret, "Intelligent heterogeneous wireless sensor networks in precision agriculture", in 2024 19th International Symposium on Wireless Communication Systems (ISWCS), IEEE, 2024, pp. 1–6.
- [17] L. García, A.-J. Garcia-Sanchez, R. Asorey-Cacheda, J. Garcia-Haro, and C.-L. Zúñiga-Cañón, "Smart air quality monitoring iot-based infrastructure for industrial environments", *Sensors*, vol. 22, no. 23, p. 9221, 2022.
- [18] Chirpstack lorawan server, https://www.chirpstack.io/, Accessed: 2024-10-29.
- [19] Influxdb, https://www.influxdata.com/, Accessed: 2024-10-29.
- [20] Mongodb, https://www.mongodb.com/es, Accessed: 2024-10-29.
- [21] Grafana, https://grafana.com/, Accessed: 2024-10-29.

## **Chatbot Integrated Holographic System for Digital Agriculture and Education**

Kübra Kara

Instituto de Investigación para la Gestión Integrada de Zonas Costeras (IGIC) Universitat Politecnica de Valencia, Valencia, Spain Email: kkubra@doctor.upv.es

Alberto Ivars-Palomares Instituto de Investigación para la Gestión Integrada de Zonas Costeras (IGIC) Universitat Politecnica de Valencia, Valencia, Spain Email: aivapal@epsg.upv.es

Abstract— The digitalization of agriculture, known as Agriculture 4.0, is revolutionizing the way agricultural products are produced, managed, and marketed. This transformation is driven by the integration of advanced technologies, such as the Internet of Things (IoT), Artificial Intelligence (AI), big data, and digital twins. Digital twins, virtual replicas of physical systems, enable farmers to simulate and optimize agricultural operations in real-time, enhancing their decision-making process. AI plays a crucial role in analyzing vast amounts of data collected from wireless sensor networks, allowing for precision agriculture and addressing global challenges, such as food security and climate change. The adoption of digital technologies also facilitates traceability and transparency in the agricultural supply chain, ensuring food safety and quality. Among the emerging technologies, holography stands out as a promising tool for intelligent agriculture. Digital holography techniques, such as computational holography, iterative holography, Fourierbased holography, Computer-Generated Holography (CGH), Deep Holography (DH), and tensor holography, offer immersive and realistic experiences for users. The integration of holography with AI and other technologies has the potential to revolutionize digital agriculture by providing real-time monitoring, management, and interaction with crops. However, the development and implementation of these technologies must be inclusive, sustainable, and focused on enhancing the quality of life for all individuals. This study presents a communication algorithm flowchart and a system proposal to elucidate the interaction between a user and a holographic system, demonstrating the potential of holography not only in digital agriculture but also in effective human learning and communication.

Keywords-Agriculture 4.0; Internet of Things (IoT); Artificial Intelligence (AI); Digital twins; Holography; Large Language Models (LLM).

#### I. INTRODUCTION

The digitalization of agriculture, also known as Agriculture 4.0, revolutionizes how agricultural products are produced, managed, and marketed. This transformation is based on the integration of advanced technologies, such as Daniel Matheus Brandão Lent Electrical Engineering Department State University of Londrina, Londrina, Brazil Email: matheus.daniel@uel.br

Jaime Lloret Instituto de Investigación para la Gestión Integrada de Zonas Costeras (IGIC) Universitat Politecnica de Valencia, Valencia, Spain Email: jlloret@dcom.upv.es

the Internet of Things (IoT), Artificial Intelligence (AI), big data, and digital twins to improve the efficiency, sustainability, and profitability of the agricultural sector [1].

Digital twins, virtual replicas of physical systems, are instrumental in agriculture. They enable farmers to simulate and optimize agricultural operations in real-time, a feature that significantly enhances their decision-making process [2]. These digital replicas collect data from Wireless Sensors Networks (WSN) distributed in fields and agricultural machinery, providing detailed information on soil conditions, weather, plant growth, and crop health [3]. By analyzing these data, farmers can make informed decisions about irrigation, fertilization, and pest control, resulting in more efficient resource use and higher productivity [4].

AI plays a crucial role in analyzing these vast amounts of data. Machine learning algorithms can predict crop yields based on historical data, optimize planting schedules, and even detect early signs of diseases or pest infestations in crops [5]. These AI-driven ideas allow precision agriculture, where water, fertilizers, and pesticides are used more effectively, reducing costs and minimizing environmental impact; this is accomplished thanks to the different technologies used, such as sensors, Global Positioning System (GPS) or the Internet of Things application (IoT) [6].

The adoption of digital technologies in agriculture not only improves crop management but also has the potential to address global challenges, such as food security and climate change [7]. This potential is underscored by a report from the United Nations Food and Agriculture Organization (FAO), which predicts a 70% increase in global food demand by 2050 [8]. Integrating technologies like artificial intelligence and big data in agriculture can help predict crop yields, optimize fertilizer and pesticide use and reduce food waste, contributing to a more sustainable future [9].

Moreover, digitalization facilitates traceability and transparency in the agricultural supply chain, ensuring the safety and quality of food that will guarantee consumer confidence [10]. Modern consumers demand detailed information about the food's origin and quality. Through blockchain and other distributed ledger technologies, it is possible to track every stage of a product's lifecycle, from

farm to table, ensuring the authenticity and quality of the final product [11].

The use of drones and IoT sensors in agriculture has also enabled more precise and real-time monitoring of fields, contributing to more effective management and reduced operational costs [12]. This technology provides farmers with critical data that can be used to improve planning and decision-making, thereby increasing productivity and reducing environmental impact, as in the case of water, where more than half of water consumption is attributed to crop irrigation. The improvement of the irrigation system will be essential to reduce consumption; this can be achieved thanks to different monitoring technologies, in this case, the monitoring of evapotranspiration [13].

AI, a key player in the digitalization of agriculture, also aids in automating repetitive tasks, such as weeding, harvesting, and sorting of produce. This practical application of AI saves time and ensures high precision and efficiency in these tasks, freeing human labor for more complex decisionmaking processes [14].

The digitalization of agriculture represents a significant opportunity to transform the agricultural sector, enhancing efficiency, sustainability, and resilience to global challenges. Integrating advanced technologies, such as digital twins and AI, promises to optimize farming operations and prepare them for a more sustainable and technologically advanced future [15].

One of these digital technologies could also be holograms. Holography is an application of optical physics, specifically the science of interference and diffraction of light and waves. It enables the creation of three-dimensional (3D) images, which, in contrast to two-dimensional (2D) images, provide enhanced detail and information due to the depth and relief characteristics of holograms. Furthermore, holography is a technology applicable to diverse fields and sectors, rendering it a multidisciplinary concept. These applications encompass medicine, engineering [16], and agriculture [17]. As with other technologies, this multidisciplinary concept has evolved into digital holography, advancing classical holography to a more sophisticated level. Digital holography possesses the capability to integrate real and virtual aspects of the world through its 3D resolution, positioning this technology within mixed reality systems. This technology can significantly enhance user engagement due to low latency, as users can interact vividly with holograms through immersion without the need of special glasses, thereby stimulating multiple human senses simultaneously through haptics, smell and taste. This type of multi-sensory technology is called mulsemedia. Moreover, an additional advantage of augmented reality technology, such as holography is its capacity for integration with multiple other technologies, a concept commonly referred to as multimodal technology or interface [18].

This research is organized as follows: Section I provides an introduction of Agriculture 4.0, emphasizing the transformation and digitalization of traditional agriculture with means of digital technologies, such as IoT, AI and specifically holography. In Section II, the related work regarding intelligent agriculture and farming, LLM, various types of digital holography are discussed. The chatbotintegrated holographic system applied to digital agriculture is being proposed in Section III along with the system's architecture and communication algorithm. Section IV presents the simulation results and evaluates the performance of different language models. Finally, Section V concludes the paper with a summary of the findings, implications of AI and holography in digital agriculture and suggestions for future research directions.

#### II. RELATED WORK

In recent years, the integration of advanced technologies into agriculture has gained significant attention. A noteworthy contribution is the intelligent agriculture system developed by Xu et al. [17]. This innovative system employs holograms to provide real-time monitoring and management of crops. By projecting holographic data related to crop growth, soil conditions, and irrigation needs, farmers are empowered to make informed decisions without the necessity of being physically present in the field. This advancement not only enhances agricultural management efficiency but also significantly reduces water resource consumption. Looking ahead, future developments in the field of intelligent agriculture could include the integration of AI for predictive analysis and the use of drones for remote monitoring. Such advancements exemplify the potential of technology to transform traditional agricultural practices into more sustainable and efficient methods, showcasing a promising future for the industry.

The importance of accessibility in digital environments is further emphasized in research by Alabi et al. [19], which focuses on visually impaired users. Their study highlights the critical role of assistive technologies, including screen readers and adaptive user interfaces, in facilitating access to information. Implementing these technologies shows that improving digital accessibility can lead to greater participation by all users, particularly those with disabilities. His research findings underscore the need for inclusive design on digital platforms and the responsibility of developers to ensure that the needs of various user groups are considered. By prioritizing accessibility, developers can create more equitable digital environments. It is intended to use this type of technology in libraries to avoid discrimination in these environments.

The development of tools aimed at improving communication for individuals with speech disabilities has also seen significant advancements. [20] Janai et al. introduced a sophisticated text-to-speech and speech-to-text conversion tool that leverages advanced natural language processing algorithms. This tool provides rapid and accurate conversion between text and voice, facilitating smoother user interactions. Its implementation across various applications has positively impacted the daily lives of those relying on such technologies, highlighting the importance of innovation in communication aids. By enhancing accessibility, these tools empower individuals with speech disabilities, fostering more significant inclusion in society.

Another significant area of development is in the realm of Large Language Models (LLM), as presented by Brown et al. [21]. Their work on models like GPT-3 represents a groundbreaking shift in how machines comprehend and generate human language. With their transformative potential, these models utilize vast datasets and sophisticated deep learning algorithms to produce coherent and contextually relevant text, which has transformed numerous applications, from content generation to academic research assistance. The capabilities of large language models extend far beyond mere Natural Language Processing (NLP); they facilitate human-computer interaction, also making technology more intuitive and user-friendly. This technological leap has opened avenues for new research, creativity, and productivity across various fields.

As previously noted by Xu et al. [17], the utilization of digital holography can be a significant asset in agriculture. The digitalization of classical holography has given rise to various holographic techniques [22]. Computational holography employs computers to generate holograms. This technique offers the advantage of adapting and manipulating holographic projections through computational parameters; however, considering the substantial data requirements of holographic projections, high bandwidth is necessary, which can be addressed through the use of algorithms and Deep Neural Networks (DNN) [23].

Iterative holography also utilizes specific algorithms, such as the Gerchberg-Saxton algorithm. As the name suggests, iterative holography continuously refines projections to reconstruct a holographic image, phase object, eliminate twin images (although beneficial in agriculture [24]), and overcome physical barriers and obstacles of the real world [25]. While this technique may result in poor image quality methods, such as the Constrained Complex Total Variation (CCTV) regularizer exist [26] to preserve image resolution.

Another method of creating holograms digitally without the use of lenses is Fourier-based holography [27], in which a reference wave enables the retrieval of phase information about any type of wave, such as light, electron, and X-ray, scattered by an obstacle, simply by using Fourier Transform to reconstruct images.

The method shown in [28] is applicable to various disciplines and provides the capability to generate Computer-Generated Holography (CGH), which leads to the next type of digital holography. The CGH method can work with incoherent light and generating high-quality holograms through the use of algorithms. Its flexibility and versatility make it suitable for Virtual Reality (VR) and Augmented Reality (AR) [29].

Next, there is Deep Holography (DH), which also utilizes DNN to execute phase aberration corrections. Additionally, DH not only gives rise to other types of digital holography, such as tensor holography, but it also addresses one of the major challenges in holograms, which is the elimination of twin images [30]. Considering the importance and relevance of twin images in agriculture, holography may be a suitable technology in digital agriculture.

Finally, very similar to CGH is tensor holography, which is the most realistic form of digital holography. Tensor holography employs DNN and Machine Learning (ML) and can also be referred to as a physics-informed DNN technique. This technique has the capacity to project real 3D volumetric images in air, making it not only the most computationally intensive digital holography technique but also the most intricate. Scholars argue that tensor holography requires 4000 pairs of RGB-depth images for training and provides the full potential of holographic video communication, thus enabling maximum user engagement and satisfaction [30][31]. Tensor holography is the type of holography that is desired to achieve revolutionizing digital agriculture. As discussed by Huang et al. [32], holography provides 6 Degrees of Freedom (DoF), involving an immersive and realistic experience in 3D movement and rotation, physically allowing users to engage with holographic projections. This again supports the relevance of holograms in digital agriculture.

While the advancements across these domains intelligent agriculture, digital accessibility, communication tools, and language models—demonstrate the profound impact of technology on society, it's important to note that they also come with potential risks and challenges. For instance, large language models can potentially perpetuate biases in the training data, leading to unintended consequences. As we continue to innovate, it is crucial to ensure that these technologies are inclusive, sustainable, and enhance the quality of life for all individuals. Embracing a collaborative approach in developing solutions can pave the way for a more equitable and accessible future.

#### III. PROPOSED SYSTEM FRAMEWORK

Large language models have proven to be a great advancement for artificial intelligence, and natural language processing. Numerous tools, such as ChatGPT, Gemini and Microsoft Copilot are already available for aiding users with faster research by interpreting multiple sources of information and answering questions. Other tools are already being studied to help health professionals to study for clinical education, practice, vascular research and scientific writing [33]. This study proposes a system to make information about different kinds of trees more accessible for people, especially biology students. It is composed of holograms, each of a different species of tree, that are connected to large language models that can answer questions about themselves. Tools for text to speech and speech to text enable the tree to talk with participants without the necessity of text input and output. The general knowledge of AI makes it ready to answer not only possible unexpected questions, but also follow up ones.

A more interactive and playful approach to learning might lead to better comprehension and knowledge retention as other studies suggest [34]. A conversation between a human and a tree allows the user to ask for clarification, or to adapt the technicality of the conversation based on the user's speech. More advanced AI are capable to, in addition

to answering questions correctly, interpreting a character, with correct usage of wordplay and even puns related to trees, which contributes to the immersion of the experience.

Figure 1 presents a chatbot-integrated holographic system capable of generating auditory and visual responses through a holographic tree to enhance user interaction. The system initiates with a spoken user input received by a microphone and transmitted in an audio stream via Wi-Fi to the Access Point (AP). The AP subsequently retransmits it to the embedded system. The user input is then converted into text through a voice-to-text translator module. This text serves as the input for the Chatbot. The chatbot's core comprises a language model that analyzes and interprets the input and generates a response based on NLP algorithms. By evaluating the probability distribution of subsequent words, the model ensures conversational coherence and contextual relevance.

The text response is transformed into speech using a textto-voice module, creating an auditory interaction. Concurrently, the system generates a holographic tree image, and the audio data is transmitted to the speakers over a Wi-Fi network from the embedded system with synchronized transmission to facilitate a conversation with the Chatbotintegrated holographic tree. The holographic tree is animated to match the output speech. The system is proposed to be viewed in an indoor environment, so when the holographic projector is not placed in the same room as in the Wireless Access point, other considerations due to wall loss must be taken into account [35][36]. Overall, the theoretical implications of this system are significant and support educational applications.

![](_page_43_Figure_4.jpeg)

Figure 2 represents a communication algorithms flowchart to illustrate the interaction between a user and a holographic system (in this case, a holographic tree). During the initialization phase, the system initiates the interaction and receives and interprets user input. Upon receiving user input, the system proceeds to the system processing stage, wherein NLP and speech recognition are utilized to process the user input, which may be a question or a command. User inputs can be as straightforward as: "Hello, can you tell me about your life as an orange tree?". The subsequent step, "Holographic tree response?", is the point at which the system evaluates whether it can generate a response to the user's input. If a response can be generated, the system proceeds to provide the user with a visual holographic response, which may be factual, educational, or procedural, based on the information retrieved from its database. If the user input is excessively complex or intricate, the system will propose that the user reformulate the prompt. Should the user decline, the system will proceed directly to the finalization stage to conclude the interaction without generating a response. If the user opts to proceed with a new prompt or input, the system will resume the cycle from the user input stage.

During the holographic tree response stage, the system creates an immersive volumetric 3D hologram or tensor hologram, as previously explained. The system's response is adaptive, dynamic, and interactive, enhancing user experience through engagement. Following this stage, the user has the option to continue the interaction or terminate it. Should the user wish to continue, the system

loops back to the input stage and maintains the flow of interaction, leading to deeper engagement. However, if the user opts to end the interaction, the system transitions to the finalization stage, concluding the cycle. This design enables the system to process various types of queries, including factual and complex questions.

![](_page_43_Figure_8.jpeg)

Figure 1. Illustration of the interaction between a user and a chatbotintegrated tensor hologram in a tree form.

Figure 2. Communication algorithms flowchart framework between a user and a holographic tree.

It is a versatile system applicable to diverse domains and topics. To contribute with immersion, text to speech tools have also evolved to better pronounce words, add intonation, pauses and stress to the voice. Neural networks are also employed to this task, as they are capable of generalizing patterns that are not always present on the learning phase [37]. Advancements were also made with multilingual models, an extremely useful feature in our use case since LLM are capable of synthesizing text in multiple languages [38]. Those same models can also simulate different voices from short audio files, to add variety for different types of trees.

By combining those three consolidated technologies with holograms, it is possible to build a prototype of the proposed system and test its performance based on some metrics: answer correctness, its preciseness and character impression.

Other metrics, such as response time were considered, but all tested models presented near-real time answers, sometimes even faster than human interaction timings, thus it was not measured. This study conducted evaluations on three pertinent and freely accessible large language models: OpenAI's ChatGPT, Google's Gemini, and Microsoft's Copilot. The same set of prompts were input across all of them to avoid inconsistencies. The models were asked to "Pretend that you are an apple tree that talks. I'm going to ask you questions about your biology and you should answer as you were this tree". Afterwards, the three models were asked a series of questions on the subject.

All the models were able to answer the questions correctly. However, only ChatGPT and Microsoft Copilot played a role of a talking tree. Google Gemini answered in the first person, as requested, but did not simulate the ludic text of a cartoonish tree.

As an example, Table 1 presents the answer to the prompt mentioned in the last paragraph. Questions about apple trees were made, from simple ones, such as "What varieties of apple trees exist", to more technical ones: "What amount of organic matter in the soil is necessary for the proper development of the apple tree and its fruit?" The AIs could answer both types of questions precisely, with no contradiction between them.

On average, the answers from the three systems were two paragraphs long. Copilot had more succinct answers and GPT the longest ones. None of them had single-line answers, even for the most direct questions, i.e., "How much water do your need to grow healthily?". Overall, the answers are the length expected from an informative conversation.

#### IV. SIMULATION RESULTS

Figure 3 represents the recorded time measurements for each module during the testing phase. The elements are ordered according to their occurrence in the proposed interaction. Initially, the Speech-to-Text (STT) delay represents the time between the conclusion of human speech and the completion of its transcription. Subsequently, the GPT Start to Answer delay indicates the interval between prompt input and the initial output from GPT.

#### TABLE I. ANSWERS TO THE INITIAL PROMPT

Model	Answer
Chat GPT	Ah, welcome beneath my branches! I am a humble apple tree, deeply rooted in the soil, stretching my limbs toward the sky. What curiosities do you bring? Ask away, and I shall answer as best as a tree can!
Gemini	Okay, I'm ready to be an apple tree! Ask me anything you'd like to know about my life.
Copilot	Alright, I'm an apple tree! My branches are adorned with lush green leaves, and I bear delicious apples. Ask away, and I'll do my best to share the secrets of my tree-ish life with you!

Third, the total time for a complete answer represents the duration required for the conclusion of GPT operation. This period is not critical for application, as the Text-to-Speech (TTS) module can commence operation as soon as the first phrase is available. Lastly, the TTS Start Delay denotes the time between text input and the initiation of audio output.

Lastly, the TTS Start Delay denotes the time between text input and the initiation of audio output. It is noteworthy that the entire system can exhibit rapid response times to the user due to the potential for task parallelization. The STT module operates concurrently with user speech, resulting in its delay being perceptible only upon the cessation of speech. Similarly, the GPT delay is noticeable only until the completion of the first phrase, at which point, in conjunction with the TTS delay, the system will begin vocalization. The GPT completes the entire answer in approximately 14 seconds, which is significantly faster than the conclusion of TTS.

![](_page_44_Figure_13.jpeg)

Figure 3. Measured response times for STT, GPT and TTS modules.

Consequently, it can be inferred that the response time delays are sufficiently low to maintain the fluidity of humanlike conversation.

As previously noted by scholars [22], transmission delay has a significant impact on user engagement and satisfaction, particularly in the context of enhancing agricultural management efficiency. This graphical representation leads to the conclusion that special attention must be directed towards delay time when incorporating AI and holographic technology in intelligent agriculture to ensure an efficiency in human-AI interactions.

#### V. CONCLUSION

In this article, we propose an interactive hologram system to present information about plants to humans. This system enables dynamic interaction between humans and machines to learn more effectively, in a ludic way. To do so, four core technologies must be combined: holograms, large language models, text-to-speech and speech-to-text tools. Holograms provide three-dimensional images from the plants studied for better visualization and comprehension of the physical aspects of the tree, in addition to better immersing the user in their learning experience.

Large language models are an efficient way of garnering information, organizing and presenting it to a human in a conversation in the most natural way. Their capabilities of synthesizing text have been used in several areas and can help swiftly accessing information combined from various sources. Tests were conducted with them to extract information about apple trees. The results show how not only do they return precise, correct answers, but also can interpret a role of a cartoonish talking tree, enhancing the experience.

Text-to-speech tools are responsible for receiving the output of the LLM and synthesizing voice. It is crucial for the user experience to have, in addition to the correct pronunciation, intonation, pauses and stress capabilities.

Neural networks have also been employed in the state-ofthe-art models due to their potential of extracting patterns from data and generalizing for unseen information. It is also possible for a single network to speak multiple languages, featured with the LLMs.

The experiment is promising, and more research is important to better implement it. For future work, more advanced features could be considered, such as interaction between two plants that have symbiotic relationships. Open source LLM models are also considered to better adjust them to the task. In addition to it, designing tree models with faces that have moving mouths could be an important step towards its implementation.

In the context of holography, a critical consideration for future development is the resolution of current technical constraints in data transmission and latency to facilitate holographic communication and the implementation of interactive holographic projections, especially in digital agriculture. This has consistently proven to be a significant factor during system evaluations when measuring STT, TTS, and GPT modules in human-AI interactions. Our future work will be focused on improving the GPT response time and testing other Conversational AI and AI Chatbots.

#### ACKNOWLEDGMENT

This work has been partially funded by the "Generalitat Valenciana" through the "Programa Investigo" project INVEST/2022/467, partially supported by Project TED2021-131040B-C31 funded by MICIU/AEI/10.13039/501100011033 and by European Union NextGenerationEU/PRTR, and partially supported by CAPES from Brazil Federal Government under project 88881.933610/2024-01

#### REFERENCES

- M. Escribà-Gelonch *et al.*, "Digital Twins in Agriculture: Orchestration and Applications," *J Agric Food Chem*, vol. 72, no. 19, pp. 10737–10752, May 2024, doi: 10.1021/acs.jafc.4c01934.
- [2] N. Peladarinos *et al.*, "Enhancing Smart Agriculture by Implementing Digital Twins: A Comprehensive Review," *Sensors*, vol. 23, no. 16, p. 7128, Aug. 2023, doi: 10.3390/s23167128.
- [3] T. Alahmad, M. Neményi, and A. Nyéki, "Applying IoT Sensors and Big Data to Improve Precision Crop Production: A Review," *Agronomy*, vol. 13, no. 10, p. 2603, Oct. 2023, doi: 10.3390/agronomy13102603.
- [4] L. A. Puntel, L. J. Thompson, and T. Mieno, "Leveraging digital agriculture for on-farm testing of technologies," *Frontiers in Agronomy*, vol. 6, Mar. 2024, doi: 10.3389/fagro.2024.1234232.
- [5] S. O. Araújo, R. S. Peres, J. C. Ramalho, F. Lidon, and J. Barata, "Machine Learning Applications in Agriculture: Current Trends, Challenges, and Future Perspectives," *Agronomy*, vol. 13, no. 12, p. 2976, Dec. 2023, doi: 10.3390/agronomy13122976.
- [6] R. P. W. A. Dilrukshi *et al.*, "Precision Agricultural Technologies for enhancing crop productivity: A way forward to Sri Lankan Agriculture," *Sri Lanka Journal of Food and Agriculture*, vol. 7, no. 2, pp. 23–42, Dec. 2021, doi: 10.4038/sljfa.v7i2.98.
- [7] O. Calicioglu, A. Flammini, S. Bracco, L. Bellù, and R. Sims, "The Future Challenges of Food and Agriculture: An Integrated Analysis of Trends and Solutions," *Sustainability*, vol. 11, no. 1, p. 222, Jan. 2019, doi: 10.3390/su11010222.
- [8] W. P. Falcon, R. L. Naylor, and N. D. Shankar, "Rethinking Global Food Demand for 2050," *Popul Dev Rev*, vol. 48, no. 4, pp. 921–957, Dec. 2022, doi: 10.1111/padr.12508.
- [9] I. Sandeepanie, "Big Data Analytics in Agriculture," pp. 1–8, Sep. 2020, doi: 10.13140/RG.2.2.25154.81604.
- [10] G. Lv et al., "Blockchain-Based Traceability for Agricultural Products: A Systematic Literature Review," *Agriculture*, vol. 13, no. 9, p. 1757, Sep. 2023, doi: 10.3390/agriculture13091757.
- [11] P. Xu, J. Lee, J. R. Barth, and R. G. Richey, "Blockchain as supply chain technology: considering transparency and security," *International Journal of Physical Distribution & Logistics Management*, vol. 51, no. 3, pp. 305–324, Jan. 2021, doi: 10.1108/JJPDLM-08-2019-0234.
- [12] A. Srivastava and J. Prakash, "Techniques, Answers, and Real-World UAV Implementations for Precision Farming," *Wirel Pers Commun*, vol. 131, pp. 2715–2746, Aug. 2023, doi: 10.1007/s11277-023-10577-z.
- [13] M. Jenkins and D. E. Block, "A Review of Methods for Data-Driven Irrigation in Modern Agricultural Systems," *Agronomy*, vol. 14, no. 7, p. 1355, Jun. 2024, doi: 10.3390/agronomy14071355.
- [14] C. Maraveas, "Incorporating Artificial Intelligence Technology in Smart Greenhouses: Current State of the Art," *Applied Sciences*, vol. 13, no. 1, p. 14, Dec. 2022, doi: 10.3390/app13010014.
- [15] K. Rijswijk *et al.*, "Digital transformation of agriculture and rural areas: A socio-cyber-physical system framework to support responsibilisation," *J Rural Stud*, vol. 85, pp. 79–90, Jul. 2021, doi: 10.1016/j.jrurstud.2021.05.003.

- [16] M. Jacquot, J. B. Carcano, R. Giust, and L. Froehly, "I learned it through the hologram," *Photoniques*, no. 119, pp. 31–35, Jun. 2023, doi: 10.1051/photon/202311931.
- [17] S. Xu *et al.*, "Development of portable user-interactive holographic information collector for agricultural product markets," *International Journal of Agricultural and Biological Engineering*, vol. 13, no. 3, pp. 143–153, 2020, doi: 10.25165/j.ijabe.20201303.5845.
- [18] M. K. Bekele, R. Pierdicca, E. Frontoni, E. S. Malinverni, and J. Gain, "Electronic display system for computational holography," *Journal on Computing and Cultural Heritage*, vol. 11, no. 2, pp. 1–36, Mar. 2018, doi: 10.1145/3145534.
- [19] A. O. Alabi and S. M. Mutula, "Digital inclusion for visually impaired students through assistive technologies in academic libraries," *Library Hi Tech News*, vol. 37, no. 2, pp. 14–17, Feb. 2020, doi: 10.1108/LHTN-11-2019-0081.
- [20] S. Janai, T. Shreekanth, M. Chandan, and A. K. Abraham, "Speech-to-Speech Conversion," *International Journal of Ambient Computing and Intelligence*, vol. 12, no. 1, pp. 184–206, 2021, doi: 10.4018/IJACI.2021010108.
- [21] T. B. Brown *et al.*, "Language Models are Few-Shot Learners," pp. 1–75, May 2020, doi: 10.48550/arXiv.2005.14165.
- [22] K. Kara, J. Lloret, O. Romero, and S. Sendra, "Performance Test of Holographic Video Communication System," in 2024 International Conference on Multimedia Computing, Networking and Applications (MCNA), IEEE, Sep. 2024, pp. 15–20. doi: 10.1109/MCNA63144.2024.10703941.
- [23] P. St-Hilaire *et al.*, "Electronic display system for computational holography," *Society of Photo-Optical Instrumentation Engineers* (*SPIE*), vol. 1212, no. Practical Holography IV, pp. 174–182, May 1990, doi: 10.1117/12.17980.
- [24] T. Shimobaba *et al.*, "Deep-Learning Computational Holography: A Review," *Frontiers in Photonics*, vol. 3, pp. 1–16, Mar. 2022, doi: 10.3389/fphot.2022.854391.
- [25] T. Latychevskaia, "Iterative phase retrieval for digital holography: tutorial," *Journal of the Optical Society of America A*, vol. 36, no. 12, pp. 31–40, Dec. 2019, doi: 10.1364/JOSAA.36.000D31.
- [26] Y. Gao and L. Cao, "Iterative projection meets sparsity regularization: towards practical single-shot quantitative phase imaging with in-line holography," *Light: Advanced Manufacturing*, vol. 4, no. 1, 2023, doi: 10.37188/lam.2023.006.
- [27] S. Mustafi and T. Latychevskaia, "Fourier Transform Holography: A Lensless Imaging Technique, Its Principles and Applications," *Photonics*, vol. 10, no. 2, pp. 1–28, Feb. 2023, doi: 10.3390/photonics10020153.

- [28] R. Zhu, L. Chen, and H. Zhang, "Computer holography using deep neural network with Fourier basis," *Opt Lett*, vol. 48, no. 9, pp. 2333– 2336, May 2023, doi: 10.1364/OL.486255.
- [29] O. Shigematsu, M. Naruse, and R. Horisaki, "Computer-generated holography with ordinary display," *Opt Lett*, vol. 49, no. 8, pp. 1876– 1879, 2024, doi: 10.1364/OL.516005.
- [30] G. Situ, "Deep holography," *Light: Advanced Manufacturing*, vol. 3, no. 2, pp. 278–300, 2022, doi: 10.37188/lam.2022.013.
- [31] Y. Huang *et al.*, "Toward Holographic Video Communications: A Promising AI-Driven Solution," *IEEE Communications Magazine*, vol. 60, no. 11, pp. 82–88, Oct. 2022, doi: 10.1109/MCOM.001.220021.
- [32] X. Huang, Y. Zhou, H. Liang, and J. Zhou, "Binocular near-eye augmented reality enabled by full utilization of functional computer generated hologram," *Opt Lasers Eng*, vol. 176, no. 108115, pp. 1–7, May 2024, doi: 10.1016/j.optlaseng.2024.108115.
- [33] F. Lareyre *et al.*, "Large language models and artificial intelligence chatbots in vascular surgery," Sep. 01, 2024, *W.B. Saunders*. doi: 10.1053/j.semvascsurg.2024.06.001.
- [34] Z. Zhang and X. Huang, "The impact of chatbots based on large language models on second language vocabulary acquisition.," *Heliyon*, vol. 10, no. 3, Feb. 2024, doi: 10.1016/j.heliyon.2024.e25370.
- [35] M. Garcia, C. Martinez, J. Tomas, and J. Lloret, "Wireless Sensors Self-Location in an Indoor WLAN Environment," in 2007 International Conference on Sensor Technologies and Applications (SENSORCOMM 2007), IEEE, Oct. 2007, pp. 146–151. doi: 10.1109/SENSORCOMM.2007.4394912.
- [36] M. Garcia-Pineda, J. Tomás, F. Boronat, and J. Lloret, "The Development of Two Systems for Indoor Wireless Sensors Selflocation," Ad Hoc Sens. Wirel. Networks, vol. 8, pp. 235–258, Dec. 2009, [Online]. Available: https://www.researchgate.net/publication/220419345\_The\_Developm ent\_of\_Two\_Systems\_for\_Indoor\_Wireless\_Sensors\_Self-location. [retrieved: October, 2024]
- [37] V. R. Reddy and K. S. Rao, "Two-stage intonation modeling using feedforward neural networks for syllable based text-to-speech synthesis," *Comput Speech Lang*, vol. 27, no. 5, pp. 1105–1126, Aug. 2013, doi: 10.1016/j.csl.2013.02.003.
- [38] E. Casanova et al., "XTTS: a Massively Multilingual Zero-Shot Textto-Speech Model," Jun. 2024, [Online]. Available: https://arxiv.org/abs/2406.04904. [retrieved: October, 2024]